

THE GEOPOLITICAL IMPACT OF CLIMATE MITIGATION POLICIES

How Hydrocarbon Exporting Rentier States and Developing Nations can Prepare for a More Sustainable Future



The Hague Centre for Strategic Studies





HCSS helps governments, non-governmental organizations and the private sector to understand the fast-changing environment and seeks to anticipate the challenges of the future with practical policy solutions and advice.

This report is from the HCSS theme GEO-ECONOMICS. Our other themes are SECURITY and GLOBAL TRENDS.

GEO-ECONOMICS

HCSS analyzes the geopolitics and economics of natural resources, including energy, minerals, water, food and land.



THE GEOPOLITICAL IMPACT OF CLIMATE MITIGATION POLICIES How Hydrocarbon Exporting Rentier States and Developing Nations can Prepare for a More Sustainable Future

The Hague Centre for Strategic Studies

ISBN/EAN: 978-94-92102-55-3

AUTHORS (all HCSS, unless indicated otherwise): Sijbren de Jong, Willem L. Auping (TU Delft), Willem Th. Oosterveld, Artur Usanov, Mercedes Abdalla, Alice van de Bovenkamp, and Christopher Frattina della Frattina.

The analyses in this report were methodologically supported by the TU Delft, Faculty of Technology, Policy and Management, Policy Analysis section, specifically, Jan H. Kwakkel and Erik Pruyt.

© 2017 *The Hague* Centre for Strategic Studies. All rights reserved. No part of this report may be reproduced and/ or published in any form by print, photo print, microfilm or any other means without previous written permission from the HCSS. All images are subject to the licenses of their respective owners.

Graphic Design Studio Maartje de Sonnaville, The Hague

The Hague Centre for Strategic Studies

Lange Voorhout 16 info@hcss.nl 2514 EE The Hague HCSS.NL The Netherlands

THE GEOPOLITICAL IMPACT OF CLIMATE MITIGATION POLICIES

How Hydrocarbon Exporting Rentier States and Developing Nations can Prepare for a More Sustainable Future

The Hague Centre for Strategic Studies

TABLE OF CONTENTS

	EXECUTIVE SUMMARY	7
	INTRODUCTION	15
1	THE OUTLOOK FOR ENERGY TRANSITION AND DECOUPLING WORLDWIDE	21
2	METHODOLOGY 2.1 Setup of the research 2.2 The energy mix model 2.3 The intra-state stability model 2.4 Scenarios	27 31 32 34 35
3	RESULTS3.1Energy mix – Oil price3.2Energy mix – Renewable energy3.3Selection price scenarios3.4Socio-political instability3.5Internal stability	37 39 43 46 52 57
4	LEVEL OF PREPAREDNESS: AN ANALYSIS OF THE OIL PRICE CRASH FROM 2014 UNTIL TODAY	61
5	 ENERGY CHOICES IN SUB-SAHARAN AFRICA 5.1 Current primary energy mix 5.2 Solar energy potential and investments 5.3 Wind energy potential and investments 5.4 Biomass 5.5 Local/decentralized energy generation 5.6 Risks of making the wrong choices 	69 71 74 75 75 76 77
6	CONCLUSIONS AND RECOMMENDATIONS	81
	BIBLIOGRAPHY	89
	ENDNOTES	103

EXECUTIVE SUMMARY

EXECUTIVE SUMMARY

At the Paris Climate Conference (COP 21) held in December 2015, 195 countries adopted the Paris Agreement – the first universal, legally binding global climate deal. The signatory parties commit themselves to a global action plan that aims to keep global warming to well below 2°C and to limit the global temperature increase to 1.5°C.

Some countries are expected to be hit harder by climate change than others, and the effects of a transition to a low carbon economy are likely to be borne unevenly. Countries richly endowed in hydrocarbons in particular stand to lose a significant part of their revenue stream if energy transition affects demand for fossil fuels. Many oil-and gas-exporting nations in North Africa, the Middle East, and the region belonging to the former Soviet Union exhibit a relatively high share of oil and gas rents as a percentage of their government budget. Such countries are often referred to as 'rentier states'. If demand for oil and gas were to decrease as a result of successful climate mitigation policies, the financial and social stability of these countries will, *ceteris paribus*, be negatively affected, especially if they fail to reform their domestic economies. This issue was acknowledged at an early stage, as signatory nations of the United Nations Framework Convention on Climate Change committed themselves to reducing the potential adverse effects of climate change mitigation measures.

Therefore, the geopolitical consequences of energy transition are an important factor, to consider when analyzing the effects of climate mitigation policies aimed at reducing greenhouse emissions. This study analyzes, both quantitatively and qualitatively, the effects of existing climate mitigation policies on countries richly endowed with hydrocarbon reserves. Specifically, we assess the conditions under which they could – in the absence of adequate reforms – result in instability and social unrest in Algeria, Azerbaijan, Egypt, Kazakhstan, Russia, Saudi Arabia and Qatar.

These countries were selected as they cover numerous important oil and gas exporting regions of the world in terms of oil and gas shipments via pipeline, as well as via liquefied natural gas (LNG). Moreover, whereas some countries saw their relative importance to the European market decline in recent years (Egypt and Libya for example), others such as Azerbaijan will see their importance grow. That said, the recent offshore natural gas discoveries in Egypt may turn this around in the future. Furthermore, these countries exhibit important differences when it comes to their governance models. Whereas, some countries are genuine autocracies (Saudi Arabia. Qatar, Kazakhstan, Azerbaijan), others such as Russia, Egypt and Libya occupy a middle-position between consolidated democracies and full-scale autocracies. Importantly, some of the selected countries have in the recent past already experienced societal uprisings (Egypt and Libya), causing it to be of interest to include them in this study. Lastly, large differences exist with respect to the financial power of the selected countries. Whereas Saudi Arabia and Qatar are home to large sovereign wealth funds, the same cannot be said about Algeria, Libya and Egypt. Countries such as Kazakhstan, Azerbaijan and Russia occupy a position in between these two groups. The different ability for these countries to financially cope with the dynamic unleashed by climate mitigation policies is therefore interesting to examine.

Although the Paris Agreement holds no specific legal obligations on the part of signatory states to fulfil their pledges, nor is there a legally binding cap on CO_2 emissions in place, it remains imperative that countries live up to their promises if the world is to seriously combat climate change. Realistically, the only way that there can be any kind of enforcement in order to make sure the national policies are implemented is to install a legally binding cap on CO_2 emissions. It is for this reason that this study explicitly looks at the effects that such a legal cap would have on the global energy mix and intra-state stability in rentier states. Following on this, the study also examines the choices made by countries in Sub-Saharan Africa with respect to their future energy mix with the aim of assessing the chances that these countries might avoid some of the choices made by industrialized nations in the past.

In this study we make use of the innovative methodology known as *scenario discovery*. Rather than to predict the future, this method is designed to perform a *what-if* type of analysis of the alternative futures that we may find ourselves in as a result of the consequences of different policy decisions that might be taken over time. In this case the selected timeframes are policy decisions taken towards 2030 and onwards to 2050. In this instance, we made use of System Dynamics (SD) models. Two different models were used: the first model represents the possible development of the global

energy mix through supply and demand interaction for oil, gas, coal, nuclear, liquid biofuels, and other renewable energy resources. Using the output from this analysis, the second model assesses the impact of economic developments on the intra-state stability of a country. As a result, one can view the analysis conducted in this study as a kind of 'stress test' of the resilience of rentier states when confronted with the effects of climate mitigation policies.

The analysis generates a number of insights. When it comes to future price levels and overall price dynamics of some of the primary sources of energy distinguished in this study a strong relation appears between the presence of a legal emission cap and the oil price. High oil prices are only possible when a lower demand for oil in capped regions is compensated by higher demand elsewhere in regions not covered by the legal cap. The opposite equally holds true: the ability to meet our CO₂ reduction targets coincides strongly with situations in which the oil price is lower. The reason is simple, as the presence of a legal cap limits the demand for oil by law. A low price of oil however also constitutes a challenge in itself. When the oil price is too competitive compared to renewable energy we witness that it is difficult to achieve a reduction of CO₂ emissions in those areas of the world that are not covered by a legal cap. The incentive to resort to the use of low-priced oil resources in non-capped regions will thus be a critical factor that could undermine and derail efforts to reduce CO, emissions. Given that a worldwide legally enforceable cap on CO2 emissions is unrealistic at this point in time it is imperative that the large-scale availability of inexpensive renewable energy sources, including liquid biofuels, is promoted. Only then, can we disincentivize the use of fossil fuels and achieve a significant reduction in CO₂ emissions in parts of the world not covered by a legal cap. Finally, instances of higher oil price levels appear to all coincide with situations in which the world fails to meet its CO₂-emissions reduction targets.

With respect to the risk of socio-political instability, the study shows that a decline in the import of fossil fuels in Europe will result in a reduction of the mutual import/ export dependencies that exist between Europe and the hydrocarbon exporting countries in our immediate neighborhood. In the absence of far-reaching reforms, and if not adequately compensated by a higher energy demand in regions that do not participate in a legal cap, the decline in oil and gas sales causes a higher risk of instability in these countries. Not every country is equally vulnerable to such a downturn however. The level of vulnerability recorded is the highest in the Middle East and North Africa, particularly in Saudi Arabia. Factors of influence in this context are the presence of a relatively large share of young people in the Saudi society that all

will need to find suitable employment in the (near) future and the enormously high level of dependence on hydrocarbon exports. Faced with declining export revenues, and a limited share of private sector employment, the government cannot simply resort to creating more public sector jobs. Another factor that may influence the risk of unrest is the discrepancy that exists between the relatively high education levels of Saudi citizens on the one hand and the harsh model of governance in the country on the other. The higher educated a population, the greater their desire for societal participation. If a government is unable to accommodate this, the risk of social unrest is likely to increase. Saudi Arabia's relatively large financial buffers may help to dampen the potential for unrest in the short term, but are unlikely to prove sufficient in the long term.

Countries that are less vulnerable to the oil price scenarios that coincide with successful climate- and energy policies are Russia and Kazakhstan. What plays a role in this context is that both countries have a relatively aged population, and the regime type corresponds well to the older generations in the country. Put differently, young people without jobs are much more likely to resort to public protests than older people who lived through the 1980s and 1990s in Soviet and post-Soviet societies. Furthermore, the Russian and Kazakh economies are more diversified compared to Saudi Arabia. Lastly, both Russia and Kazakhstan are among the countries with the best geographical possibilities for building up an economy based on the export of liquid biofuels that could offset some of the loss in export revenues based on their centrality between the key-markets in Europe, India and China.

A major conclusion is that a lack of attention for the second-order effects of energy transition in the form of heightened instability, in particular in the Middle-East, runs the risk of creating adverse geopolitical consequences in the form of state failure and collapse comparable to the 2011 Arab uprisings. Specific factors that are indicative of a heightened vulnerability are: a high share of resource rents in the GDP in combination with limited financial reserves; a high national debt as a percentage of the GDP or a rapid increase in a state's national debt; a young and/or relatively fast growing population (Middle East and North Africa), a relatively high share of youth unemployment and uncertainty about the continuation of the existing political leadership; a combination of high subsidies and a high domestic energy demand; and the position in international financial markets (risk-aggravating factors are a low credit rating, and difficulties to borrow in international markets). A second major conclusion is that recognizing that countries face this potential time bomb this could undermine their willingness to push for a more ambitious climate policy. Put differently, expect them to try and undermine other countries' efforts at combating climate change.

At the same time, we should anticipate that the incentive for countries to again resort to fossil fuels will increase when prices will decrease as a result of a legal cap on CO₂ emissions in some parts of the world. Countries not covered by the cap may be tempted to increase their use of fossil fuels and countries under the cap system may be incentivized to withdraw. Internationally, this will require lobbying at the highest level to motivate countries to take part in a legal carbon cap system and continued investment in renewable energy with the aim of cutting costs. For that purpose it is of major importance to create *prospects of gain*: both the Netherlands and the EU should be able to demonstrate clearly the economic benefits of decoupling and of a transition towards renewable energy.

A different but related challenge manifests itself in developing countries that face a choice between investing in renewable energy networks or to rely on fossil fuels for their economic growth. A major issue in this context is the need to reduce the high use of biomass in Sub-Saharan Africa, owing to its hazardous health effects and the problems associated with indirect land use change. In the long term it is crucial that a lower use of biomass does not result in a higher reliance on fossil fuel resources. This means that, particularly when viewed through the lens of ensuring coherence in Dutch international policies, the government will have to determine how to deal with public (financial) bilateral contributions to the exploration of new hydrocarbon finds in developing countries.

In the absence of wide-reaching reforms in the countries analyzed in this study, many will face scenarios in which they will struggle to deal with domestic unrest. The biggest risk thereof will manifest itself in the Middle East and North Africa. A silver lining considering energy transition is that, as the name of the phenomenon implies, this is not something that will happen overnight. This means that there are still several decades for countries heavily dependent on the export of fossil fuels to take adequate measures to reduce their vulnerability. This however requires a fundamental reassessment of the nature of the relationship that we have with this part of the world. The crux of the matter is that whereas western nations have an interest in stabilizing the MENA region, the countries in this part of the world have a key interest in reducing their vulnerability with an eye on future regime survival. When reforming the domestic economies of these countries, the focus should be on addressing the factors labelled as critical in this study.

Switching from fossil fuel-based power generation to one based on renewable energy can, on the one hand, mean a significant cost reduction by eliminating wasteful energy subsidies, whilst reducing the burning of otherwise valuable export resources on the

other. In the bilateral ties and sectoral dialogues between the Netherlands and countries in the MENA region – the Gulf countries in particular – it is important to stress that doing so represents a significant cost-cutting measure at a time when export revenues are under pressure, and would also allow for the diversification of the economy through the buildup of a domestic renewable energy sector that can compete internationally. For the Netherlands this represents a business opportunity to showcase technological advances in the areas of renewable energy, energy efficiency and clean-tech.

In convincing countries that it is in their interest to do so it is useful to take note of the Chinese experience. The Chinese leadership's growing concern about the impact of climate change - partly fed by growing popular discontent about the level of air pollution - have pushed the country in the direction of an aggressive policy that promotes renewable energy. Today, China is the world leader in domestic investment in renewable energy and associated low-emissions energy sectors, and increasingly, the country has been investing in renewable energy projects abroad with the aim of expanding its renewable sector. Gulf countries with significant sovereign wealth funds, such as Saudi Arabia and Qatar, can learn a thing or two from how China built up such a strong domestic renewable energy sector, followed by a sizeable overseas expansion strategy in frontier markets. If they succeed in building up a domestic renewable energy sector, it would be beneficial as a follow-on step to invest in the energy transition in hydrocarbon exporting countries that are financially less wellendowed and in relative proximity to the European mainland, such as Algeria and Egypt, so that the export of electricity becomes feasible. This would help rentier states around Europe to lessen their dependence on commodity sales to Europe. More importantly, it would dampen the need for domestic energy subsidies and reduce the negative effects that climate mitigation policies will have on these countries. For Saudi Arabia and Qatar, it would result in an ability to collect a return on their investments and to generate additional export revenues allowing these countries to hedge against lower oil prices and diminished income from hydrocarbon exports in the future. For Europe, this would mean that - provided the required infrastructure is financed - it can tap into an additional source of renewable energy that is located in relative proximity, whilst the risk of large-scale instability in our immediate neighborhood is reduced.

INTRODUCTION

INTRODUCTION

At the Paris Climate Conference (COP 21) held in December 2015, 195 countries adopted the Paris Agreement: the first universal, legally binding global climate deal. Under the terms set out in the accord, countries commit themselves to a global action plan that aims to keep global warming to well below 2°C and to limit the increase to 1.5°C, as doing so would significantly reduce the risks and impacts of climate change. With respect to reducing greenhouse gas emissions, the agreement strives to reach a peak in global emissions as soon as possible and to undertake rapid reductions thereafter.¹

The Paris Agreement introduced the mandatory submission of Nationally Determined Contributions (NDCs) at the nation state level. Although not enough in-and-of themselves, the NDCs, which are to be updated every five years, are a step in the right direction. Every five years governments are to report to each other on the progress achieved and to set out their new ambitions in accordance with the latest scientific and technological innovations. After 55 countries, representing 55% of global emissions, had ratified the agreement, it entered into force on 4 November 2016.²

The Netherlands works within its own domestic agreements and frameworks to decarbonize its economy and ensure a greener future. The 2013 Agreement for Sustainable Growth aims to raise the Netherlands' share of renewable energy in the energy mix to 14% by 2020 and 16% by 2023.³ The 'Energy Report' (Dutch: Energierapport) that was released in January 2016 identifies CO₂ reduction as its primary vehicle to steere the transition in the period 2023-2050. The report states a desire on the part of the Dutch government to reduce greenhouse gas emissions by 80-95% by 2050 in accordance with the agreements made at the European level.⁴ In 2015, the share of renewable energy in the Netherlands' energy mix rose from 5.5 to 5.8%. By 2023, it is expected that this share will have increased to 15.8%.⁵ By 2050 the government is clear in its ambitions. In the 'Energy Agenda' (Dutch: Energie-

agenda) that was published in December 2016 the government states that by 2050 virtually all electricity should be generated sustainably; buildings will be heated primarily through geothermal energy and electricity; it expects companies to have adapted their production processes accordingly; households will no longer use gas as a primary source for cooking; and that virtually all cars driving in the Netherlands will be powered through electricity.⁶

For some this level of ambition is not yet enough. In June 2015 a Dutch court ruled that the Dutch state must do more to reduce greenhouse gas emissions. In particular, the government should aim to reduce CO_2 emissions by at least 25% by 2020.⁷ The case was brought before the court by Dutch climate organization 'Urgenda'. The Dutch government has since appealed against the verdict.⁸

The impacts of climate change are unlikely to be felt evenly across the world. Some countries are expected to be hit harder than others. By the same token, the effects of a transition to a low carbon economy are expected to be borne unevenly. Although the planet as a whole stands to benefit from climate mitigation in environmental terms, countries richly endowed in hydrocarbon reserves potentially stand to lose a significant part of their revenue stream if the energy transition affects demand. Many oil- and gas exporting nations in North Africa, the Middle East, and the region belonging to the former Soviet Union exhibit a relatively high share of oil and gas rents as a percentage of their government budget. Such countries are often referred to as 'rentier states'.⁹ If demand for oil and gas were to decrease as a result of successful climate mitigation policies, the financial and social stability of these countries would, *ceteris paribus*, be negatively affected, especially if they fail to reform their domestic economies. This issue was acknowledged at an early stage, as signatory nations of the United Nations Framework Convention on Climate Change (UNFCCC) committed themselves to reducing the potential adverse effects.¹⁰

Therefore, the geopolitical consequences of energy transition are an important factor, to consider when analyzing the effects of climate mitigation policies aimed at reducing greenhouse emissions. This study analyzes, both quantitatively and qualitatively, the effects of existing climate mitigation policies on countries richly endowed with hydrocarbon reserves. Specifically, we assess the conditions under which climate mitigation policies could – in the absence of adequate reforms – result in instability and social unrest in Algeria, Azerbaijan, Egypt, Kazakhstan, Russia, Saudi Arabia and Qatar. These countries were selected as they cover numerous important oil and gas exporting regions of the world in terms of oil and gas shipments via pipeline, as well

as via liquefied natural gas (LNG). Moreover, whereas some countries saw their relative importance to the European market decline in recent years (Egypt and Libya for example), others such as Azerbaijan will see their importance grow. That said, the recent offshore natural gas discoveries in Egypt may turn this around in the future. Furthermore, these countries exhibit important differences when it comes to their governance models. Whereas, some countries are genuine autocracies (Saudi Arabia, Qatar, Kazakhstan, Azerbaijan), others such as Russia, Egypt and Libya occupy a middle-position between consolidated democracies and full-scale autocracies. Importantly, some of the selected countries have in the recent past already experienced societal uprisings (Egypt and Libya), causing it to be of interest to include them in this study. Lastly, large differences exist with respect to the financial power of the selected countries. Whereas Saudi Arabia and Qatar are home to large sovereign wealth funds, the same cannot be said about Algeria, Libya and Egypt. Countries such as Kazakhstan, Azerbaijan and Russia occupy a position in between these two groups. The different ability for these countries to financially cope with the dynamic unleashed by climate mitigation policies is therefore interesting to examine.

In this context, it should be pointed out that although the submission of NDCs is a mandatory process under the Paris Agreement, there is no legal obligation on the part of signatory states to actually fulfil these national pledges – this remains a national prerogative – nor is there a legally binding cap on CO_2 emissions in place. This means that it remains to be seen how much of the national pledges will ultimately be put into practice. Looking towards the future however, and if the world is serious about limiting the increase in global temperatures to $1.5^{\circ}C$ it is imperative that countries live up to their promises. Realistically, the only way that there can be any kind of enforcement in order to make sure the national policies are implemented, is to install a legally binding cap on CO_2 emissions. It is for this reason that this study explicitly looks at the effects that such a legal cap would have on the global energy mix and intra-state stability in rentier states.

Following on this, it is important to look not only at countries richly endowed in hydrocarbons, but also at developing nations that are yet to make the choice between opting for a future based on the exploration and utilization of fossil fuels or, instead, pursuing an energy mix that largely consists of renewable energy sources and hence avoids some of the choices made by industrialized nations in the past. For this reason, the report explicitly looks at the choices made by countries in Sub-Saharan Africa with respect to their future energy mix.

This report consists of six chapters. Chapter 1 kicks off by providing an overview of the outlook for energy transition worldwide. Chapter 2 details the methodology used in this study by explaining the various models that underpin the quantitative analysis. Chapter 3 discusses the initial results based on the model runs conducted with the energy mix- and the intra-state stability model. Chapter 4 delves into the consequences of the oil price crash witnessed between 2014 and today and the responses of hydrocarbon exporting nations. Chapter 5 looks to the future by examining the energy choices made, and about to be made, in Sub-Saharan Africa and the consequences they carry. Finally, Chapter 6 concludes with a number of implications and policy recommendations.

1 THE OUTLOOK FOR ENERGY TRANSITION AND DECOUPLING WORLDWIDE

1 THE OUTLOOK FOR ENERGY TRANSITION AND DECOUPLING WORLDWIDE

There are three main ways to reduce greenhouse gases. First, the overall energy efficiency of the economy can be raised, which corresponds to a 'decoupling' of economic growth and energy demand. Many advanced economies in the world have reached the stage of 'decoupled' economic growth, whereby a growth in GDP is not matched by a corresponding increase in energy demand. In other words, the economy can grow without necessarily increasing the pressure on our planet in terms of a greater demand for energy resources. A second way to reduce greenhouse gas emissions is to increase the share of renewable energy sources in the energy mix. Finally, one can capture and sequestrate greenhouse gas emissions. The first two strategies in particular are aimed at structurally changing the global energy system.¹¹

Currently (i.e., in 2014, the latest year for which data is available), renewable energy sources account for 14% of the global total primary energy supply (TPES; Figure 1).¹² By far the largest renewable energy source, representing two thirds of total renewable energy supply presently consists of solid biofuels or charcoal. While hydro energy provides 18% of all renewable energy globally, all other renewable energy sources including wind, solar, and modern biofuels represent only 15% or 2% of the TPES.



FIGURE 1. WORLD RENEWABLE ENERGY SUPPLY BY SOURCE IN 2014. SOURCE: IEA.

The high share of solid biofuels is explained by their widespread use (i.e., mainly for home heating and cooking) in developing countries, particularly in Africa and Asia.¹³ Their use is often associated with negative health effects, and many countries aim to replace them by providing the population with access to electricity and modern fuels. This is primarily why the share of renewable energy sources in non-OECD countries has declined significantly since 1971 (Figure 2). An opposite trend can be witnessed in OECD countries: the share of renewables roughly doubled to 10% between the 1970s and 2014, with significant growth only beginning around 2005. Globally, renewables also added approximately 1.5 percentage points to their share in the TPES in the last few years, yet their share has not changed much since 1971, illustrating the challenges of energy transition on the global level.



FIGURE 2. SHARE OF RENEWABLE ENERGY SOURCES IN TPES, 1971-2014. SOURCE: IEA.

Over the last 25 years, the most rapidly growing renewable sources of energy in OECD countries were liquid biofuels and solar PV, which grew by 44% per year on average while TPES increased at the rate of 0.6% per annum over the same period.¹⁴

More recent data shows that 2015 was a boom year for renewable energy. Renewable electricity capacity increased by a record 153 gigawatts, mainly due to additions in wind and solar photovoltaics (PV) installations. For the first time, renewable energy accounted for more than half of the total net annual additions to globally installed power capacity.¹⁵ Global investment in renewable energy reached US\$ 286 billion, which also set a new record, exceeding the previous one that had been set in 2011.¹⁶ It was also the first time when investment in modern renewables (excluding large hydro) in developing countries was higher than that in developed countries. China alone

accounted for 40% of the global renewable energy expansion. The largest recipients of new investments were solar and wind energy, which received US\$ 161 and US\$ 110 billion respectively. All other renewable energy sources attracted a combined investment total of US\$ 15 billion. What is particulary noteworthy is that this boom in renewable energy had taken place at the time when low fossil fuel prices made traditional hydrocarbons significantly more cost competitive than before. The reasons for this rapid expansion of renewable energy capacity can therefore be attributed to significant cost reductions of some renewable energy technologies on the one hand, in particular solar PV, and to favorable government policies on the other.¹⁷

There is little doubt that renewable energy will continue to expand in the future. The Paris Agreement on climate change signed by 195 countries in December 2015 provides an additional impetus for such an expansion. Yet, the speed of this growth is highly uncertain and depends on a multitude of factors including the pace of technological progress, prices of competing energy sources, government policies, and public acceptance.

The IEA World Energy Outlook 2016's 'New Policies Scenario', which reflects pledges made by countries under the Paris Agreement, expects that renewable energy sources will grow on average by 2.6% per annum in OECD countries and 2.1% per annum in non-OECD countries between 2014-2040, with solar and wind energy providing most of the growth. This will ensure that the share of renewables will reach 20% of total energy demand by 2040 in both OECD and non-OECD countries. Another, more ambitious scenario, called the '450 Scenario', illustrating a pathway to keep global warming below 2 °C, calls for much more rapid growth of renewables, with their share growing to 32% by 2040 (33% in OECD and 31% in non-OECD countries, respectively).¹⁸ The reality will probably lie somewhere in between these projections.

2 METHODOLOGY

2.1	Setup of the research	31
2.2	The energy mix model	32
2.3	The intra-state stability model	34
2.4	Scenarios	35

2 METHODOLOGY

In this study we make use of the innovative methodology known as *scenario discovery*.¹⁹ The goal of using this approach is to explore how a system may evolve and what the causes of possibly interesting futures are. Put differently, this method explores what the origins may be of the different kinds of worlds that we may find ourselves in as a result of different decisions taken over time. Due to the complexity of the problem that is analyzed, it is necessary to use simulation models. In this case, we made use of System Dynamics (SD) models.²⁰ Naturally, these models represent a simplified version of reality. As a result, they are not suitable for predicting the future. Rather, they are designed to perform an analysis of the consequences of different combinations of assumptions and uncertainties. In other words, these models allow for a *what-if* type of analysis of the consequences of predefined choices, parameters or vantage points – in this case with respect to climate mitigation policies.²¹

The full range of assumptions that one can have about a system can be interpreted as a 'mental model'.²² The simulation models used in this study are thus an internally consistent representation of existing mental models. In cases where there are differences between existing mental models, or where experts agree that they cannot know how the system functions, one can speak about 'deep uncertainty'.²³ By varying the input used for the simulation models it is possible to generate a large bandwidth of 'plausible futures', or possible future worlds that are a result of the input (data, assumptions, uncertainties, parameters, etc.) used. Within the set of plausible scenarios it is then possible to make a selection based on the desirability of each respective scenario. Translated to climate mitigation, assuming we are serious in our desire to combat climate change, this implies that those scenarios in which CO_2 emission reduction targets are met are considered desirable. The uncertainties that lie at the basis of the set of plausible futures thus represent the causes of the 'undesirable' behavior uncovered in the model analysis. More extreme, yet plausible futures, can thus also be used as a way of testing the robustness of existing or policy decisions under

consideration.²⁴ The models in this study are thus used to generate an internally consistent narrative for potentially interesting futures or end states. The models as such act as a tool for generating a narrative, whereby the uncertainty analysis is performed to generate an as large as possible, yet plausible, bandwidth within these narratives. In other words, the analysis would cover the whole range of plausible end states that we can find ourselves in as a result of the set of assumptions, and uncertainties that exist and the decisions that are taken at the outset with respect to climate mitigation.

In this study we chose to combine quantitative and qualitative research methods for a variety of reasons. The complexity of the global energy system and of intra-state stability implies that it is difficult, if not impossible, to come to a complete understanding of the future development of these systems. The use of simulation models, SD models²⁵ in this case, enables us to make assumptions and uncertainties explicit and to explore the consequences of these assumptions and uncertainties on the future behavior of the system.

Complexity is defined as the existence of feedback loops between the various elements of the system. Instances of such complexity in the global energy system are numerous. The most common examples are that, *ceteris paribus*, whereas a higher price leads to greater supply and a lower demand, a greater supply and a lower demand result in a lower price. Such a system thus consists of two balancing feedback loops. Other examples can be found in the substitution behavior between different primary energy sources, where a high price of energy source A can induce a greater use of energy source B, provided they share similar characteristics. Within a state system an economic downturn can induce instability, the onset of which in turn hampers further economic development. This is thus an example of a self-reinforcing feedback loop. Policies, such as those aimed at climate mitigation, can create additional feedback loops, thus rendering the system even more complex.

Deep uncertainty exists when "analysts either do not know, or decision makers do not agree about: what the (most) suitable conceptual model is to describe the relationship between the most important factors of influence in the long term; what the probability distributions are that are used to portray the uncertainty of important variables and what the parametric values are in the mathematic representation of these conceptual models, and/or, how the desirability of different outcomes should be valued."²⁶ Examples of deep uncertainty that are of relevance in the context of this study include the cost development of both fossil as well as renewable resources and the development of life expectancy or education levels within societies.

2.1 Setup of the research

In this study we use two different simulation models (Figure 3). The first one represents the development of the global energy mix through the interaction between supply and demand. It includes different scenarios for the development of prices for oil, gas, biofuels, and other renewable energy sources. These price scenarios subsequently act as the input for a simulation model on intra-state stability that can be parametrized for the different oil and gas exporting countries around Europe. In this way, we can assess the impact of different developments in the global energy market on the stability of these countries. Examples of such developments are changes in climate mitigation policies. As a result, one can view the analysis conducted in this study as a kind of 'stress test' of the resilience of rentier states when confronted with the effects of climate mitigation policies.



FIGURE 3: THE RESEARCH DESIGN

The energy model is regionalized for four different parts of the world (i.e., Europe and adjacent regions, North America, the Far East, and the rest of the world). Every region has a single price level for each individual energy source. As a consequence, Europe, the Middle East and North Africa, and the countries belonging to the former Soviet Union form one region. These regions are also characterized by the existence of interconnected pipeline networks. Neither in the Far East nor in the fourth region, 'the rest of the world', is there such an interconnected pipeline network. Rather, natural gas is transported via maritime routes, primarily in the form of liquefied natural gas (LNG). The energy model distinguishes between six primary sources of energy: oil, gas, coal, nuclear, liquid biofuels, and other renewable energy resources. By using this model we were able to generate eleven different price scenarios that could be used as input for the intra-state stability model.

The intra-state stability model assesses the impact of economic developments on the satisfaction of the general population in a country. This is also known as the 'greed' aspect of intra-state stability. The other side of the equation, the more ideologically driven cases of instability, is known as the 'grievance' aspect.²⁷ In this model we explicitly look at the influence of energy prices on resource rents, which in turn influence a country's economic development and carry consequences for the labor market and people's purchasing power. Looking at these aspects in combination with demographic developments, the level of democratization and developments in a country's security apparatus enables us to study changes in social-political stability.

2.2 The energy mix model

The energy mix model consists of six different sub-models.²⁸ These sub-models pertain to energy demand, supply, the costs of supply, commodity prices, commodity trade and CO₂ emissions. Figure 4 shows how these various sub-models interact.



FIGURE 4: THE ENERGY MARKET MODEL

In addition to the above, there are two other important drivers of system behavior that are external to this model. These are exogenous economic growth, primarily driven by non-energy related changes in the productivity, and endogenous effects due to pricing of primary energy sources. Examples of relations between the various sub-models are: an increase in demand causing higher prices; higher prices causing lower demand; a greater extraction capacity for fossil fuels causing higher costs; the total use of different energy resources ultimately determining the level of CO₂ emissions; the regional difference between supply and demand determining the necessity of imports, or the possibility of exports.

An important uncertainty in this particular model is the question whether or not economic costs are incurred when a legal cap on CO_2 emissions obliges countries to render fossil extraction capacity inactive, in other words, to keep some of their fossil fuel resources in the ground. The principal thought behind this is that the potential scarcity that arises as a result could lead to situations where – periodically – there is

not enough energy available for the desired level of economic activity. An example thereof is the practice of 'load shedding', or denying districts or industries access to the electricity grid during certain hours of the day.

2.3 The intra-state stability model

The intra-state stability model used in this study is parametrized for seven different countries. These are Algeria, Azerbaijan, Egypt, Kazakhstan, Russia, Saudi Arabia and Qatar (Figure 5). All seven of these countries have, in different shapes and forms, important energy ties to the EU.



FIGURE 5: THE INTRA-STATE STABILITY MODEL

In this country stability model we make use of five different sub-models. These submodels pertain to natural resource extraction, economics, demographics, institutions, and instability. The most important exogenous drivers of the model are a country's autonomous economic growth – and derived from the energy mix model – the prices for the different resources produced. Examples of the types of relations between the various sub-models are: income from natural resource production feeding economic growth; more economic growth enabling a greater availability of jobs; economic development possibly reducing the number of children born per family, thereby changing the demographic composition of a country; and increases in instability in turn hampering economic growth.

2.4 Scenarios

In order to select plausible future scenarios from the large number of potentially interesting price dynamics, we made use of eleven selection criteria. A criterion consists of different demands that can be placed on for example prices or CO_2 emissions, or the maximum possible share that a particular energy source can reach within the energy mix under the conditions and parameters that were defined at the outset (Figure 6).

SCENARI O NAME	I NDEX
1 - Highest average oil price in cap GE	1
2 - Highest average oil price in cap GE and NA	2
3 - Highest average oil price in cap GE, NA and FE	3
4 - Highest gas share with cap	4
5 - Highest biofuels share	5
6 - Lowest biofuels share	6
7 - Highest other renewables share	7
8 - Lowest other renewables share	8
9 - Highest oil share with cap	9
10 - Lowest global CO2 emissions	10
11 - Highest average oil price with global emission target	11

FIGURE 6: SCENARIOS. GE= GREATER EUROPE. NA= NORTH AMERICA. FE= FAR EAST.

Scenarios 1-3 assume the highest possible average oil price in the event that there is a legally binding cap on CO_2 emissions in place for the different regions in the world. This should be interpreted as follows: when there is an emission cap in a particular region, this shows the impact of a forced reduction in CO_2 emissions in roughly a quarter of the global economy. Since it cannot be ruled out up front that in the future the entire world will have to move towards a legally binding emissions cap, it is relevant to incorporate this possibility into the study. Also, we know that oil and gas exporting countries benefit from higher energy prices. These three scenarios therefore

illustrate the situation in which a legal cap would have the least negative impact on the stability of large hydrocarbon exporting countries. The other scenarios (4-11) consider the circumstances in which a particular type of energy either gains the highest or the lowest share in the energy mix. This matters because it allows us to assess the effects that climate mitigation policies have on stimulating particular forms of energy.

3 RESULTS

3.1	Energy mix – Oil price	39
3.2	Energy mix – Renewable energy	43
3.3	Selection price scenarios	46
3.4	Socio-political instability	52
3.5	Internal stability	57

3 RESULTS

The results of the study can be subdivided into the changes recorded in the energy mix as a result of climate mitigation policies on the one hand, and the impact on intrastate stability on the other. This chapter consists of five sections. Section 3.1 looks at the impact of climate mitigation policies on the oil price, particularly focusing on the dynamics that could be witnessed if and when limits are imposed on carbon emissions. Section 3.2 assesses the development of renewable energy sources within the energy mix in the event of legally imposed caps on CO₂ emissions. Section 3.3 discusses the selection of plausible future scenarios. Sections 3.4 and 3.5 look at the impact of the various price scenarios on the socio-political stability and internal stability of the hydrocarbon exporting countries examined in this study.

3.1 Energy mix – Oil price

Figure 7 displays the full range of oil price dynamics that are the result of two thousand simulations ('runs') with the energy mix model. The distribution of the runs in 2050 is shown on the right hand side.



ORE 7: UIL PRICE DYNAMICS

Oil price dynamics

A number of conclusions can be drawn. Roughly speaking we can distinguish three types of dynamics. For the first type, an initial dip is followed by either a constant oil price or one that rises slowly from the level recorded in 2010. The second type is a scenario whereby the price continues to drop after the first dip and experiences high levels of volatility between the price level of 2010. The third type is a scenario whereby the price drops after the first dip, has high levels of volatility and subsequently lies significantly below the price level of 2010. The initial dip that is visible in all scenarios is explained by the advent of the US shale revolution.

A second conclusion is that in roughly three quarters of the simulations the price level in 2050 lies below that of the current pricing environment. A quarter ends up having a price level roughly equal to the one witnessed today. Given that this cannot be explained in a probabilistic manner, we will now examine the causes of these different scenario classes.

Oil price dynamics with legal emissions cap

Figure 8 shows the same model runs as those displayed on the previous page (Figure 7). The color corresponds to the geographical area in which a legal emissions cap would be in place.



FIGURE 8: OIL PRICE DYNAMICS WITH LEGAL EMISSIONS CAPS IN DIFFERENT REGIONS

Figure 8 shows that if no region has a legal emission cap in place, oil prices either rise slightly or remain around the level recorded in 2010. As soon as one or more regions install a legal emissions cap, oil prices start to decline. If more than one region has a legal emissions cap, this, on average, results in lower oil prices compared to the initial price level. Only in a limited number of runs was it possible to avoid having a lower oil price compared to the 2010 level. A legal emissions cap in one particular region coincides with a higher oil price only if other regions are able to compensate for the lower demand in the 'cap region' by having a higher energy demand themselves. In these instances, the level of CO₂ emissions will ultimately remain high. A large degree of oil price volatility can be discerned in all cap-situations. We do however see that the average price level of volatile scenarios is lower in the case of a legal emission cap.

Oil price dynamics when CO2 targets are met

We also examined the effects on the oil price when proposed emission reduction targets are met. In Figure 9, it is clearly visible that climate goals are not met in situations where the oil price is higher in 2050 than it was in 2010. When the oil price remains roughly at the level recorded in 2010, meeting climate goals is possible only if the energy mix consists of a sufficient share of renewable energy. This conclusion is illustrated by the differences between Figure 9 and Figure 10 below.



FIGURE 9: SCENARIOS WHEREBY CO2 REDUCTION TARGETS ARE MET AND NOT MET



FIGURE 10: SCENARIOS WHEREBY THE SHARE OF BIOFUELS IN THE ENERGY MIX IS EITHER HIGH OR LOW

Figure 9 displays the full bandwidth of scenarios in which CO_2 reduction targets are met as well as instances in which they are not met. Figure 10 shows the full bandwidth of scenarios in which it is possible to have either a high or a low share of biofuels in the energy mix. When comparing Figure 9 and Figure 10, note the blue colored lines at the bottom of both figures. The blue lines in Figure 9 indicating scenarios where CO_2 reduction targets are met match with the blue colored runs in Figure 10, in which the share of biofuels in the energy mix is high. This shows that unless we actively promote this type of fuel we are unlikely to meet our emission reduction targets.

Conclusions on the oil price

On the basis of the analysis in Section 3.1, a number of conclusions can be drawn. First, there are three types of scenarios for future oil price dynamics: a relatively constant or mildly rising oil price; a sharply declining oil price followed by a constant price level; or an oil price with a relatively large degree of volatility. The bifurcation in the oil price between high and low levels is caused by the presence of a legal cap on CO_2 emissions.

The difference between the scenarios shown in this study and those regularly published by institutions such as the International Energy Agency or the US Energy Information Administration is the manner in which the oil price can decline and the absence of an oil price that rises continuously. From historical data we know that the oil price can decline rapidly in a relatively short amount of time and remain at a low price level afterwards for a prolonged period. This is precisely what has happened between June 2014 and today. The low oil prices are caused by the growth in substitution possibilities at higher prices than those that preceded it and the increase in extraction capacity for unconventional energy resources. There is a strong relation between the presence of a legal emissions cap and the oil price. In particular, high oil prices are only possible when reduced demand in a cap region (or cap regions) is compensated by higher energy demand in non-cap regions. At the same time, the ability to meet CO₂ reduction targets coincides strongly with lower oil price levels. The reason is simple, as the presence of a legal cap limits the demand for oil by law. The price pressure exerted by renewables is however less pronounced in these situations.

3.2 Energy mix – Renewable energy

Next to seeing what could happen to the dynamics of the oil price as a result of climate- and energy policies, it is interesting to assess the possible developments for liquid biofuels and other renewable sources of energy. In doing so, this paragraph specifically looks at the bandwidth of the share of renewable energy sources in the energy mix under different policy conditions. In addition, we assess the effects brought about by Dutch, European, and other climate- and energy policies worldwide. We thereby assume, by way of a thought experiment, that the rest of the world will not, as of yet, apply a stringent climate- and energy policy of its own. The reason for doing so is that it is important to know the effects of non-participation of countries in a legal cap system on global CO_2 emissions. Finally, this section turns to discussing the impact of a lessened dependence on fossil fuels on the security of supply.



Legal emission cap in Europe and adjacent regions, and North America

--- Legal emission cap in Europe and adjacent regions, North America, and the Far East

FIGURE 11: RELATIVE SHARE OF RENEWABLE ENERGY IN THE ENERGY MIX IN EUROPE AND ADJACENT REGIONS
Figure 11 shows the total bandwidth within which the share of biofuels can oscillate in Europe and its adjacent regions. Here again, we can see a strong link with the presence of a legal emissions cap. Note that in this context the number of regions in which such a cap exists is in fact not relevant as the only instance in which higher shares of biofuels are not possible occurs when there is no such a cap in the first place. It should however be stressed that high shares of biofuels always coincide with a relatively strong decoupling between economic growth and energy demand. In this particular situation (i.e., the presence of a legal cap and strong decoupling), the total energy demand in Europe decreases. We know that such a situation would put the prices of all the different energy types under pressure. That said, the current level of maturity of biofuels is not up to the level required to firmly affect other types of fuel. Therefore, it will be practically impossible before 2050 to reach a production level of biofuels comparable to the level of oil and gas in the current energy mix.

Impact of a legal cap on CO₂ emissions worldwide

It is realistic to assume that the presence of a legal cap in large parts of the world will not necessarily lead to an overall reduction in energy demand elsewhere (i.e., in those regions not covered by the cap) (Figure 12). This is due to the dampening effect that a legal cap has on energy prices, which in turn reduces the necessity to achieve greater decoupling of economic growth and energy demand in the rest of the world.



FIGURE 12: CO₂-EMISSIONS IN THE REST OF THE WORLD (MILLION TONS PER YEAR)

Conversely, renewable energy sources will also become cheaper, causing their share in the energy mix to increase, which in turn can reduce the total volume of CO_2 emissions in the world. In our analysis we see that this latter effect indeed can cause CO_2 emissions to be reduced slightly. However, for cases in which the oil price is too competitive for renewable energy we see that no reduction of CO_2 emissions relative to the situation where no legal cap exists in any of the regions considered in this study. The incentive to resort to the use of low-priced oil resources in non-capped regions will thus be a critical factor that could undermine and derail efforts to reduce CO_2 emissions.

This leads us to conclude that in order to achieve a reduction in CO_2 emissions in the rest of the world, the large-scale availability of inexpensive renewable energy sources, including liquid biofuels, is sorely needed. Only then, can we disincentivize the use of fossil fuels and achieve a significant reduction in CO_2 emissions in parts of the world not covered by a legal cap. The best possible result however will be achieved when a legally enforceable cap exists in all regions worldwide.

Security of supply

A reduction in the share of fossil fuels in our energy mix does not automatically mean that security of supply improves solely due to a lessened dependence on autocratic regimes from which hydrocarbon resources are imported. A number of reasons exist as to why this is not the case.

First, the Netherlands and the rest of Western Europe are relatively densely populated. The consequence of this is that land is relatively expensive, while societal resistance against resource extraction or production is sizeable. For example, witness the protests against the natural gas exploration in the Dutch province of Groningen. The same goes for the production of renewable energy resources and the installation of windmills at sea. Given that the production of oil and gas within the Netherlands is relatively expensive compared to other parts of the world, it can be assumed that if oil and gas demand were to subside significantly, the first production centers that close are those in relatively high cost areas. That in turn means that the relative level of dependency on these resources increases.

For the same reason it should not be assumed that when there is a strong increase in the share of biofuels and other renewable energy resources in the energy mix, these resources will necessarily be produced in the Netherlands or elsewhere in Europe. Local resistance against windmills on land and the relatively high cost of placing these installations at sea are a factor to consider. What is more, the available space is limited due to the presence of waterways and fishing grounds. At the same time, a higher level of solar irradiance in areas near the equator also means that these parts of the world have a greater potential for renewable energy.

Furthermore, even if the overall expenditure in the economy that is devoted to energy resources decreases as a result of decoupling, the economy as a whole will still remain dependent on energy. If, as a result of decoupling, local energy production becomes less cost-effective and thus a greater share of our energy is imported from abroad, the share of the economy dependent on energy imports as well as our overall level of dependency will increase. This further underlines the importance of for example the strengthening of the EU internal energy market and initiatives such as the Energy Union which aim to foster a greater integration of electricity and gas pipeline infrastructure.

3.3 Selection price scenarios

As described in Section 2.4, out of approximately two thousand model runs we selected those scenarios that would enable us to perform a 'stress test' for extreme, yet plausible energy price scenarios. Given that we know from previous research²⁹ that traditional oil and gas producing countries are most vulnerable to scenarios with a declining oil price, scenarios 1-3 look at cases with the highest possible oil price on average under conditions whereby a legally enforceable cap on emissions is in place (Figure 13).

SCENAR IO NAM E	NDEX
1 - Highestaverage oilprice in cap GE	1
2 -Highestaverage oilprice in cap GE and NA	2
3 -Highestaverage oilprice in cap GE,NA and FE	3
4 -Highestgas share with cap	4
5 -Highestbiofuels share	5
6 - Low estbiofiels share	6
7 - Highestother menew ables share	7
8 - Low est other renew ables share	8
9 -Highestoilshare with cap	9
10 -LowestgbbalCO2 em issions	10
11 -Highestaverage oilprice with globalem ission target	11

FIGURE 13: OVERVIEW OF SCENARIOS

Scenarios 4-9 look at the highest or lowest possible contribution to the energy mix of different types of energy in 2050. In ascending order, they look at the highest relative contribution of natural gas – often referred to as a transition fuel towards a low carbon economy; the highest and lowest share of liquid biofuels; the highest and lowest share of other renewable energy resources; and finally the highest share of oil in the energy mix under conditions whereby a legally enforceable emissions cap is in place. Scenario 10 is used to see with which energy mix we can reach the lowest level of CO_2 emissions worldwide, and which energy prices correspond to such as scenario. Finally, scenario 11 looks at the highest average oil price in the event that a global emission target is in place.

Oil price scenarios

The selected oil price scenarios contain three scenarios in which the oil price is usually higher than recorded in 2010 (scenarios 1, 2, and 3, Figure 14). The other scenarios (4-11) show a relatively lower price level compared to 2010. What is noteworthy about scenario 11, the scenario that depicts the highest possible oil price when global CO_2 -emissions reduction targets are met, is that the oil price is continuously below the price level of 2010. In the long term, the price level is even significantly lower.



FIGURE 14: SELECTED OIL PRICE SCENARIOS

Higher oil price levels (scenarios 1-3) therefore all coincide with a situation whereby the world fails to meet its CO_2 -emissions reduction targets. In these scenarios, demand from the rest of the world compensates for a reduction in oil demand in capped regions.

Other price scenarios

Both the oil price and the biofuels scenarios demonstrate a high degree of volatility (Figure 16 and Figure 15). This has to do with the fact that these fuels can be 'stocked'. When the market suffers from overcapacity, stocks will thus accumulate. Although the development of large scale batteries is moving forward gradually and natural gas can be stored in depleted gas fields, the options for storing electricity and natural gas remain limited compared to oil and biofuels.







FIGURE 16: BIOFUELS PRICE SCENARIOS

For example, excess capacity of natural gas is often flared, whereas renewable energy sources that connect to the electricity grid are simply switched off in the case of overcapacity.



FIGURE 17: GAS PRICE SCNARIOS



FIGURE 18: OTHER RENEWABLES PRICE SCENARIOS

In this study we assumed that it will be difficult for the price of biofuels to dive significantly below the price of oil. This makes sense when taking into account that biofuels and oil have a comparable functionality. Given that biofuels have an advantage over oil in that they produce only a fraction of the latter's CO₂ emissions, the minimal price of biofuels thus becomes the price of oil minus any costs associated with *carbon pricing*.

Scenarios in perspective

In Figure 19 we have plotted the eleven scenarios used in this study amidst the entire oil price range that was the result of the 2000 model simulations. What is visible is that the eleven scenarios cover virtually all types of energy price dynamics. In other words, the selected scenarios cover the available 'scenario space' relatively well, especially at the beginning of the period (between 2015 and 2020) and at the end (2050).



FIGURE 19: THE ELEVEN SCENARIOS IN PERSPECTIVE (OIL PRICE)

When looking at the global level of CO_2 emissions it is even more clearly visible that the eleven scenarios used in this study cover the available scenario space well (Figure 20).



FIGURE 20: THE ELEVEN SCENARIOS IN PERSPECTIVE (CO, EMISSIONS)

Only a few cases in which we witnessed a very high level of CO_2 emissions (see the grey lines at the top of Figure 20) – coinciding with a wholly unsuccessful climate- and energy policy – lie outside the bandwidth of the eleven scenarios used in this study. However, given that this study departs from investigating the impact of (successful) climate- and energy policy under the influence of a legally binding CO_2 emissions cap, it is not problematic that this small number of model runs lies outside of the bandwidth of the selected price scenarios.

Looking at the development of energy demand in Europe gives a comparable image as witnessed for CO₂ emissions (Figure 21).



FIGURE 21: ENERGY DEMAND IN EUROPE AND ADJACENT REGIONS

Only those cases with a relatively high energy demand – meaning that the level of decoupling is comparatively small – are not adequately covered by the scenarios used in this study. However, given that these situations strongly correlate with failed climate- and energy policies on reducing CO_2 emissions, again, it is not considered problematic that these cases lie outside of the bandwidth of the eleven price scenarios that were used.

Renewable energy

We now turn to the effects that an increase in energy demand has on the energy mix. In the second phase of this study we use the eleven selected price scenarios as input for a model that assesses the socio-political and intra-state stability within 'rentier states' around Europe.

Before we do so, however, we can already draw a number of conclusions. It is sensible to assume that the costs for biofuels and other renewable energy sources continue to drop when the Netherlands and Europe focus more strongly on utilizing renewable energy for power generation. As a result, biofuels and other renewable energy sources can be expected to become more competitive in the energy mix, which in turn is likely to lead to their greater use in parts of the world where renewable energy sources may not be explicitly supported at the governmental level. This could strengthen the economic position of the Netherlands and Europe by raising their exports of liquid biofuels as well as of relevant knowledge and technologies. The potential for renewable energy sources, both with respect to liquid biofuels and other sources of renewable energy, in the countries studied in this report is highly uncertain. This caveat notwithstanding, the potential for solar energy in these countries is generally estimated to be high, which can be explained by the relatively greater level of experience of these countries with this type of energy compared to liquid biofuels. Also, the geographical location of the rentier states of the Middle East and North Africa contributes to a greater estimate of the potential for solar power.

This uncertainty set aside, the utilization of renewable energy sources can possibly compensate some of the expected loss of revenue due to a diminished return from the export of hydrocarbons. It can do so in two possible ways. On the one hand, renewable energy sources such as liquid biofuels can be stocked and exported, thus possibly contributing to generating valuable export revenues that can offset some of the losses incurred. On the other hand, the utilization of solar power to generate electricity could replace the burning of (expensive) fossil fuel resources for power generation and reduce the need for employing wasteful fuel subsidies, thus freeing up cash that can be used for other purposes.

3.4 Socio-political instability

Previous HCSS research demonstrated that a decline in fossil fuel prices can have destabilizing effects on the internal stability of countries that rely on resource rents to a large degree.³⁰ A number of different factors can be considered 'critical' in this regard. They can, therefore, potentially serve as *early warning* indicators of socio-political unrest.

Critical factors

First, the natural resources sector, particularly the oil sector, comprises a relatively large part of the economy in most rentier states (measured as % of oil and gas rents in GDP). For the countries under investigation in this study, oil and gas rents account

for an even greater percentage of government revenues compared to their contribution to GDP (Table 1).

VARIABLE	OIL RENTS AS % OF GDP (2012)	GAS RENTS AS A % OF GDP (2012)	OIL/GAS RENTS AS % OF STATE BUDGET	EU SHARE OF OIL EXPORTS	EU SHARE OF GAS EXPORTS	REGIME TYPE
Russia	13,9%	2.3%	52% (2012)	79% (2012)	81% (2012)	Anocracy
Algeria	17,1%	5,9%	60% (2013)	72% (2013)	90% (2013)	Anocracy
Egypt	8,0%	3,2%	N/A	56% (2013)	7% (2013)	Anocracy
Qatar	12,1%	12,5%	60% (2012)	N/A	30% (2012)	Autocracy
Saudi Arabia	45,8%	2,4%	90% ⁸⁶ (2012)	15% (2012)	No exports	Autocracy
Kazakhstan	24,9%	2,2%	N/A	56% (2012)	Net importer	Autocracy
Libya	52,3%	2,1%	96% (2012)	71% (2012)	100% (2012)	Anocracy

TABLE 1: RESOURCE RENTS, EU SHARE OF EXPORTS AND REGIME TYPE³¹

Second, all of the countries we analyze, with the exception of Qatar and Kazakhstan, saw their 'fiscal break even oil price' rise dramatically (Table 2). Especially countries in the Middle East and North Africa started spending a lot more in the wake of the Arab uprisings of 2011 with the aim of preventing similar unrest in their own nations.

	ALGERIA		EGYPT		SAUDI- ARABIA		QATAR		KAZAKHSTAN		RUSSIA		AZERBAIJAN	
	2010	2015	2010	2015	2010	2015	2010	2015	2010	2015	2010	2015	2010	2015
Fiscal Break even oil Price (US\$)	85	130	-	-	69.5	106	76	60	68.9	68	116.7	110	33.4	95
Sovereign Wealth Fund (% GDP)	29%	23%	No SWF	Planne d	41%	89%	55%	120%	25%	68%	10%	10%	22%	46%
Foreign Exchange Reserves (% GDP)	96%	74%	16%	9%	79%	90%	24%	32%	18%	13%	27%	19%	14%	7%

TABLE 2: FISCAL BREAK EVEN OIL PRICES, SOVEREIGN WEALTH FUNDS AND FOREIGN EXCHANGE RESERVES³²

Third, large financial buffers enable countries to compensate for losses in export revenues. Financial buffers usually consist of foreign exchange reserves or sovereign wealth funds (SWFs; Table 2). What matters in this context is the relative, rather than the absolute, size of the financial buffers compared to the size of the overall economy. Countries such as Qatar and Saudi Arabia clearly stand out in this regard compared to the other countries in this study. In the long term however, Saudi Arabia does have a problem given the high level of defense expenditure (modernization and the costs associated with the war in Yemen), coupled with the increase in social spending in the wake of the Arab uprisings.³³ Russia, Azerbaijan, and Algeria occupy a notably weaker position compared to Saudi Arabia and Qatar when it comes to the relative size of

their financial reserves. The national currencies in Russia, Kazakhstan and Azerbaijan all depreciated considerably in 2014-2016 as a result of the oil price decline. However, such a devaluation does somewhat compensate for a reduction in oil export revenues (see also Chapter 4), as the national budget is denoted in the national currency. Furthermore, a devaluation makes domestic products cheaper, which is beneficial for these countries' export positions. Unfortunately, for many of the countries in this study, little else beyond hydrocarbons is exported. The economic slowdown in China and Russia acts as a knock-on blow for the economies in Kazakhstan and Azerbaijan.

Fourth, in the Middle East in particular, the presence of migrant workers is often used as a buffer mechanism. In case of a recession, work visa for labor migrants are often revoked leading to a slowdown in the rise of unemployment among the local population. Such policies thus have a dampening effect on the potential for social unrest.

A fifth factor that matters is regime type, or the kind of 'polity' by which a country can be classified (see Table 1). Research by the Center for Systemic Peace has demonstrated that full autocracies and democracies are more stable over time than so-called 'anocracies'.³⁴ Anocracies represent partial democracies whose populations enjoy partial liberties. Countries in Northwestern Europe are all democracies without exception. Saudi-Arabia is an example of a full autocracy, whereas countries such as Russia or Algeria can be qualified as anocracies. The reason for the relative instability of anocracies derives from the fact that partial liberties of the population often lead to a desire for a greater level of liberty in society. What also matters in this context is the level of education of the population. In general, the higher the level of education, the greater the desire for a democratic form of governance.

	ALGE	RIA	EG	YPT	SAUDI-A	RABIA	QAT	AR	KAZAH N		RU	SSIA	AZERB	AIJAN
	2010	2015	2010	2015	2010	2015	2010	2015	2010		2010	2015	2010	2015
Youth Unemployment (% age 15-24)	22.2	25.2 (2014)	26.3	38.9 (2013)	29.8	28.7 (2013)	1.3	1.5 (2013)	6	4.5	16.8	14.5 (2013)	14.6	14.8 (2013)
Change in food prices (% yr on yr)	1.3.	1.4	22.3	8.3.	4	2.2	-0.81	0.9	2.0	3.6	õ	20	-1	5.83

A sixth and seventh factor relate to changes in employment and purchasing power, particularly youth unemployment and a rise in basic food prices (Table 3).

TABLE 3: YOUTH UNEMPLOYMENT AND CHANGES IN FOOD PRICES³⁵

Table 3 shows that food prices rose by 20% in Russia, 8.3% in Egypt, and 5.83% in Azerbaijan between 2010 and 2015. The combination of economic decline, currency devaluation and rising food prices undermines the purchasing power of the population. All countries experienced a relatively high level of youth unemployment, with the exception of Qatar and Kazakhstan. The issue is particularly salient in Algeria, Egypt, and Saudi Arabia. These countries are thus under tremendous pressure to provide an increasing share of young people with jobs.³⁶

Potential for renewable energy

The potential for renewable energy can be seen as a way to compensate losses in income resulting from a drop in oil and gas export revenues. However, it has to be stated up front that the data (Table 4) with respect to the potential for renewable energy is highly uncertain.

	POTENTIAL (B	POTENTIAL (BBTU/YEAR)						
COUNTRY	BIOFUELS	OTHER RENEWABLES						
Algeria	56472	597482282						
Azerbaijan	44866	3592291						
Egypt	356	4347958						
Kazakhstan	8973 or 30000	123196741						
Russia	59821	112123592						
Saudi Arabia	0	3329900964						
Qatar	0	15901						

TABLE 4: RENEWABLE ENERGY POTENTIAL PER COUNTRY³⁷

As can be seen from Table 4, the estimates for the potential for biofuels are often very low, at times even lower than the actual production levels of biofuels despite that being theoretically impossible. This discrepancy is due to the fact that liquid biofuels are not well developed, and there is also some level of ambiguity with respect to what *should* and what *should* not be included when speaking about biofuels. That being said, it should be assumed that the total future potential for biofuels is larger than what is indicated in Table 4.

The potential of biofuels notwithstanding, it should be pointed out that their production is not uncontroversial. Biodiesel production in OECD countries increased by 92% from 2000 to 2010.³⁸ Initially, biofuels were considered an environmentally friendly practice and biofuel programs were heavily subsidized. As a result, large swaths of agricultural

land were transformed into land for biofuel crop production. The boom contributed to a steep increase in the non-food demand for crops like soy and corn, and between 2007-2009 biofuels accounted for 20% of the global use of sugar cane, 9% for both oilseeds and coarse grains, and 4% for sugar beet.³⁹

However, in the years that followed it became apparent that the promotion of biofuels had a negative effect on food security. Often biofuels are produced from crops that are diverted from existing food production. The effect thereof is that the resultant gap in the food supply is partly filled with the expansion of cropland worldwide, which in turn likely leads to greater carbon emissions through indirect land use change.⁴⁰ The IMF and the World Bank assessed that 70%-75% of the increase in food prices during the global food crises of 2008 could be attributed to a growing demand for biofuels.⁴¹ As a consequence, many countries are now reversing their biofuel policies.

For the estimates with respect to solar and wind energy the opposite is true. Often the estimates for the potential are very high. An example of this can be seen in the estimate for Saudi Arabia. Sometimes this is because such estimates are done on the basis of the total solar irradiance on land, as in the case of Saudi Arabia. As a result, this potential should be judged as theoretical, as it does not take into account any practical limitations that may exist on land. Therefore, it should be assumed that the *real* potential is much lower than indicated. Lastly, this overestimation of the potential can also partly be attributed to the relatively high level of maturity found in solar and wind energy technologies. Finally, it should be noted that it is unlikely that Egypt, Saudi Arabia, and Qatar will be able to export electricity to Western Europe as long as this is not cheaper and cannot be transported. This limitation however does not count for the other countries or for biofuels.

3.5 Internal stability

For all countries in the study we subsequently examined whether the 11 price scenarios and two reference scenarios lead to an improvement in internal stability compared to the situation in which the oil price remains at the 2010 level (see Figure 22 for the example of Igeria).



All countries demonstrate a uniform result: only in scenarios with a relatively high oil price did the level of stability increase on average. We know that the three scenarios in which the oil price is relatively high all coincide with a failing of global climate- and energy policies. All other cases in which climate policies are successful lead to a relative increase in instability in all countries investigated. Based on this part of the analysis, however, we cannot (yet) say which countries suffer relatively more or relatively less from this development.







FIGURE 24: INTERNAL STABILITY IN SAUDI ARABIA

Figure 23 and Figure 24 show the possible developments with respect to internal stability in Russia and Saudi Arabia. The blue lines correspond to a scenario whereby the oil price is relatively high (scenario 1), the green lines indicate a scenario with a relatively low oil price on average (scenario 10). It is clearly visible that for both countries the blue as well as the green lines have the same initial bandwidth. This can be attributed to the 'shale effect'⁴² and the fact that the impact of climate- and energy policy is unlikely to be significant at the beginning of the period under consideration. When comparing Figure 23 with Figure 24 it can be seen that there are scenarios in which, irrespective of the oil price development, Saudi Arabia could become very unstable. It is however clear that a higher oil price level always leads to less instability. Following on this, it is clear that the relative order of magnitude of instability in Saudi Arabia is larger than in Russia. The primary reason for this difference lies in the fact

that Saudi Arabia has an even less diversified economy than Russia does, has a relatively young population and already suffers from high levels of youth unemployment. Furthermore, Saudi Arabia has limited possibilities to develop an economy based on the export of renewable resources, given that connecting the Saudi electricity network to nearby countries is not realistic at this point in time due to the presence of several violent conflicts in its immediate vicinity.

Economic development

The one-sidedness of the Saudi Arabian economy is clearly visible when we compare the possible development of Russia's GDP with that of Saudi Arabia (see Figure 25 and Figure 26).



FIGURE 25: DEVELOPMENT OF RUSSIAN GDP



FIGURE 26: DEVELOPMENT OF SAUDI GDP

Economic development in the case of Saudi Arabia remains very strongly linked to the price of oil. This means that unless the Saudi economy is significantly diversified, it will be very difficult for Saudi Arabia to experience economic growth under a scenario in which global climate mitigation policies are successful.

Conclusions on social-political instability

The most important conclusion of the above analysis on the influence of climate mitigation policies on social-political stability in rentier states around Europe is that countries in the Middle East, Saudi Arabia in particular, are highly vulnerable to these policies. That said, this is a development that will only become noticeable in the long term. Factors that play a major role in this context are the relatively young population and the enormously high level of dependence on the export revenues of oil and/or gas. It is however not unimaginable that a large discrepancy between the polity classification and the governance model in a country on the one hand, and the level of education of the population on the other also play an important role. The relatively large financial buffers may help to dampen the potential for unrest in the short term, yet are unlikely to prove sufficient in the long term.

We find that Russia and Kazakhstan are less vulnerable to oil price scenarios that coincide with successful climate- and energy policies. What plays a role in this context is that both countries have a relatively ageing population, and the regime type corresponds well to the older generations in the country. Also the Russian and Kazakh economies are more diversified compared to Saudi Arabia. Finally, Russia and Kazakhstan are among the countries with the best geographical possibilities for building up an economy based on the export of liquid biofuels. That said, the critique against the use of biofuels formulated in Section 3.4 should be borne in mind.

Although the effects of energy transition will primarily manifest themselves in the long term, some of the possible dynamics could already be witnessed over the past two years when average crude oil price declined from US\$ 108 to US\$ 47 per barrel (bbl)⁴³, or by 56%, and have since largely stayed within the range of US\$ 40-US\$ 60/bbl. However, arguably brought about by the US shale oil revolution, which shifted the demand and supply balance in the oil market in favor of consumers, the energy transition should also put a significant downward pressure on fossil fuels including oil products. In this respect studying the impact of low oil prices on oil exporters and the measures they have taken to adjust to new conditions provides an important insight on how these countries might react to the energy transition. For that reason Section 4 takes a closer look at which policy measures major oil- and gas exporting countries have taken to cope with the adverse economic conditions.

4 LEVEL OF PREPAREDNESS: AN ANALYSIS OF THE OIL PRICE CRASH FROM 2014 UNTIL TODAY

4 LEVEL OF PREPAREDNESS: AN ANALYSIS OF THE OIL PRICE CRASH FROM 2014 UNTIL TODAY

Between 2010 and 2015 oil production in the US expanded by 70% and the country became the largest oil producer in the world. OPEC countries as a whole and Saudi Arabia in particular also increased their production in order to defend their market share and to force shale producers out of business. The oil price shock has had a significant negative effect on oil-exporting countries: their export and tax revenues declined substantially. Economic growth has also slowed and in some cases turned into a recession. All oil- and gas-exporting countries have thus been forced to undertake urgent policy measures to counter the negative economic circumstances.

There are some limitations to the analysis conducted in this chapter. Energy transition is expected to be a relatively slow-moving process giving countries much more time to adjust compared to the current oil price shock which took place within a much shorter time frame. Correspondingly, the responses of many countries have focused mainly on short-term measures rather than on long-term plans.

In addition to the countries for which we performed a quantitative analysis (Algeria, Azerbaijan, Egypt, Iran, Kazakhstan, Qatar, Russia and Saudi Arabia, see Chapter 3), we looked qualitatively at Nigeria, Turkmenistan and Venezuela.

Coping measures can be organized into several broad categories, for example:

Macroeconomic and fiscal responses are typically short-term measures, while structural and political measures often have a much longer-term horizon. In some cases government actions can be assigned to several categories simultaneously. For example, the privatization of state-owned companies can be seen as a way to raise government revenues, but it also represents a structural measure intended to increase the effectiveness of such companies and to attract private investment. All oil-exporting countries have experienced a negative impact of the oil price decline on their government budgetary revenues. Between 2013 and 2015, in all countries with the exception of Egypt and Iran the general government revenues dropped as a percentage of GDP.⁴⁴ In the case of Saudi Arabia the drop was 16% of GDP. In Algeria, Azerbaijan and Kazakhstan the decline was also very significant – above 5% of GDP (Figure 27).



FIGURE 27. CHANGE IN GENERAL GOVERNMENT REVENUE BETWEEN 2013 AND 2015, IN PERCENTAGE POINTS OF GFP. SOURCE: IMF.

To deal with the budgetary problems, all countries had to undertake various measures to bridge the gap between spending and revenues. The most common type of response was a reduction of government spending, in particular, on long-term investment projects (Azerbaijan, Kazakhstan, Russia, Saudi Arabia), energy and water subsidies (Egypt, Qatar, Nigeria, and Saudi Arabia), and public salaries (Algeria, Russia, and Saudi Arabia). Some countries also raised taxes (Algeria, Azerbaijan, Iran, Russia, and Turkmenistan).

Most oil-exporting countries accumulated substantial financial reserves before the crisis (in the form of foreign exchange and sovereign wealth funds) and used them to pay for public sector expenditure when economic circumstances worsened. In some countries these reserves allowed them to pursue the expansion of public spending programs, in particular in Qatar, where infrastructure spending was protected from cuts ahead of the FIFA 2022 World Cup.⁴⁵ However, the size of financial buffers varies significantly. In countries such as Kazakhstan, Qatar, and Turkmenistan they can finance 20-30 years of projected budget deficits but in other states such as Venezuela and Egypt they are essentially empty (see also Section 3.4).

One of the macroeconomic adjustment measures that oil-exporting countries used was currency depreciation. Currency depreciation mitigates the impact of oil price declines on budgetary revenue and increases economic competitiveness, yet this comes at the expense of increased inflation and associated declines in the purchasing power of the population. Some countries initially resisted the depreciation (Azerbaijan, Egypt, Kazakhstan, and Nigeria), often using restrictions on currency transactions, but were eventually forced to devalue under the pressure of financial markets. Only countries with very large reserves, Saudi Arabia and Qatar, were able to keep their exchange rate at the same level as in 2014 at the cost of large depletions of their reserves. Another macroeconomic step that many countries had to undertake was an increase in their interest rates – to reduce capital outflows and soften currency depreciation. Devaluation and interest spikes have created a significant risk for the stability of the banking system. To address this challenge, some countries recapitalized banks (Azerbaijan, Algeria, and Russia), purchased non-performing loans from banks (Azerbaijan, and Kazakhstan) and increased the supply of liquidity.

In terms of structural measures, the most ambitious plan was adopted by Saudi Arabia: Vision 2030.⁴⁶ It presents a strategy for long-term economic diversification, channeling more money into high-tech sectors and raising the share of Saudis employed in the private sector. The goal of the project is to move Saudi Arabia from its current position of the world's 19th largest economy into the top 15.47 Instrumental within the Saudi plans is the partial privatization of the national oil company Saudi Aramco and its transformation into a global industrial conglomerate. The ownership of Saudi Aramco and other state-owned companies is transferred to the Public Investment Fund, which will then become the largest sovereign wealth fund in the world. The plans further sets out to diversify the economy away from its reliance on hydrocarbons and localize the oil and gas sector, ultimately raising the share of non-oil exports in non-oil GDP from 16% to 50% by 2030. In doing so, the Saudi government set itself a goal of generating 9.5 gigawatts worth of renewable energy, seeking to localize a significant portion of the renewable energy value chain in the Saudi economy. Energy prices are in the long term to be brought on to the free market level and labor force participation, particularly for women, is supposed to increase.⁴⁸ To facilitate this goal, the government will issue tenders for a large-scale solar and wind power program worth between US\$ 30 and US\$ 50 billion. That said, renewable energy is not the only game in town. The Kingdom is also said to be in the early stages of feasibility and design proposals for the country's first commercial nuclear power stations, with a capacity of 2.8 gigawatts.⁴⁹ Although grand in its ambitions, Vision 2030 is scant on details. To achieve its goals, Saudi Arabia would have to significantly open up to global trade, investment, allow foreign visitors and implement international codes of conduct including greater transparency and secular laws. Here Saudi deputy crown prince Muhammad Bin Salman, the man behind Vision 2030, is up against powerful vested forces within Saudi society, including the Saudi clerics and the business interests in his extended family.⁵⁰ Time will tell if the prince is able to follow through on his bold promises.

Kazakhstan has also launched structural reforms. On a more practical level, Doing Business reports produced by the World Bank⁵¹ show that it was Kazakhstan that made most significant improvements in the overall business climate for entrepreneurship over the last two-three years, while the regulatory conditions worsened in Qatar, Saudi Arabia and Venezuela.

Has the oil price decline had an impact on the internal political situation and foreign policies of the oil-exporting countries? This question is difficult to answer even with an in-depth analysis of a particular country because these areas are affected by a myriad of factors – most of which not directly related to the price of oil. Some broader measures of democratic governance suggest that there is no uniform trend and countries moved in opposite directions. For example, World Governance Indicators (voice and accountability) indicate that Nigeria has improved its score mainly due to the generally fair and peaceful presidential election in March 2015. However, the situation deteriorated in Azerbaijan, Qatar, and Venezuela. Azerbaijan became increasingly intolerant against dissent resorting to the imprisonment of journalists and activists. In Venezuela, after the opposition won the parliamentary election in December 2015, President Nicolás Maduro declared a state of emergency in May 2016 and the political situation remains tense.

Another indicator, military expenditure, in particular as a share of GDP or government expenditure, also provides an insight on leadership's priorities. Azerbaijan, Russia, and Saudi Arabia show a substantial increase with regard to this indicator between 2013 and 2015.⁵² It is not a coincidence that these three were involved in various conflicts in 2015-2016: Nagorno-Karabakh, Syria, and Yemen, correspondingly. In all three countries these military engagements were popular domestically. In Venezuela military expenditure declined significantly owning to the collapsing economy.

Conclusions

This brief review shows that all oil-exporting countries are negatively affected by a decline in oil prices and have had to adopt various fiscal and monetary policies to

adjust to a new reality. Their record on structural policies that would promote longterm non-oil economic growth has been limited, although some announced ambitious plans for the future. The impact of the decline in oil prices on oil-exporting countries' politics is quite uncertain. However, it is possible to say that in a relatively short-term perspective (over the last two years), it did not lead to their foreign policy restraint and may, in fact, have had the opposite effect.

5 ENERGY CHOICES IN SUB-SAHARAN AFRICA

5.1	Current primary energy mix	71
5.2	Solar energy potential and investments	74
5.3	Wind energy potential and investments	75
5.4	Biomass	75
5.5	Local/decentralized energy generation	76
5.6	Risks of making the wrong choices	77

5 ENERGY CHOICES IN SUB-SAHARAN AFRICA

When looking beyond the current oil price environment and its effects, one has to focus in particular on the energy choices made by developing countries. After all, for energy- and climate policies to bear fruit it will ultimately matter a great deal as to whether or not developing nations make similar choices as industrialized nations have in the past. For this reason, this chapter takes an in-depth look at the energy choices made by Sub-Saharan African countries and the plans these countries have for the future.



5.1 Current primary energy mix

To meet their energy needs, many economies in Sub-Saharan African countries rely to a large extent on biomass (Figure 28).

FIGURE 28: SUB-SAHARAN AFRICA PRIMARY ENERGY MIX BY SUB-REGION, 2012. SOURCE: IEA, AFRICA ENERGY OUTLOOK 2014.

However, the share of oil and gas in domestic energy consumption varies greatly across the continent. Nigeria for example relies on oil for 9% of its total energy consumption. For natural gas this figure stands at 5%.⁵³ In Mali, fossil fuels make up around 21% of the primary energy mix.⁵⁴ Ethiopia relies on fossil fuels for 8.3% of its domestic energy consumption.⁵⁵ These shares are significantly higher in other countries. Senegal for example relies on oil for 40% of its domestic energy consumption.⁵⁶ In terms of its domestic energy consumption, Gambia is 30% reliant on oil products.⁵⁷ South Africa heavily relies on coal, which comprises 68% of its domestic energy mix.⁵⁸

Hydroelectric power, although representing only 1% of the overall energy mix in Sub-Saharan Africa, is nonetheless a significant source of energy in several countries. Ethiopia (86%), Central African Republic (56.8%) and Equatorial Guinea (77%) all have high shares of hydropower in their primary energy mix. A new wave of investments in hydro energy is expected in Sub-Saharan Africa, slated to triple the existing capacity by 2040.⁵⁹ Although renewable and a clean and reliable source of energy with low operating costs, there are some drawbacks to hydroelectric power. First and foremost, it is expensive to build, as it requires significant up front infrastructural investments. It is true that this can be offset by the low operating costs afterwards, yet for Sub-Saharan African countries the initial capital costs can be prohibitive. Second, the damming of waterways is known to be disruptive to ecosystems. Third, the onset of dry spells could seriously hamper the ability of a hydroelectric power plant to deliver enough power. Power and energy costs are specifically identified with the amount of water that is accessible and the onset of a drought can have a highly disruptive effect.⁶⁰

In some countries, geothermal energy acts as a significant source of energy. One report concludes that "geothermal will play a small role, but relative to total known energy capacity on the continent, it will have significant volume, reaching 28 terawatthours by 2040".⁶¹ At present, it is an important source of energy in Ethiopia, Sudan, and Kenya, where in 2014, it made up over half of these countries electricity output.⁶²

Biomass however is by far the most used source of energy across the continent, making up 49% of the energy mix in Angola, 62% in Gabon, 70% in Gambia, and 78% in Mali. Other countries with a relatively high share of biomass in their primary energy mix are Central African Republic, Kenya, Mali, Mozambique, Nigeria, Senegal, and Sudan. These countries are largely representative of the composition of the energy mix throughout Sub-Saharan Africa (Figure 29).



FIGURE 29: TOTAL PRIMARY ENERGY DEMAND FOR ENERGY SOURCES IN AFRICA. SOURCE: IEA, 2010, FROM IRENA-BIOMASS POTENTIAL IN AFRICA.

A 2014 IEA report states that 60% of the energy needs in sub-Sahara Africa are met through bioenergy.⁶³ What is more, "consumption in the region continues to increase and its growth since 2000 has been greater than that of all other fuels combined."⁶⁴ At the same time, the overall share of bioenergy is expected to decline to 47% by 2040.⁶⁵

All countries in Sub-Saharan Africa invest in energy infrastructure in one way or another. In order to further lessen its dependence on oil in its energy mix, Nigeria recently introduced a regulation that makes it mandatory for companies to acquire a minimum percentage of electricity from renewable energy resources.⁶⁶ In view of global climate regulations, much funding is geared towards renewable energy. However, not all initiatives are ultimately successful, and nor does investment into renewables automatically equate greening the economy, as, for instance, biomass can also be an important source of local pollution and hazardous to human health when used indoors.

Ethiopia is particularly ambitious, wanting "to become a regional renewable energy hub in East Africa."⁶⁷ Kenya is also making significant investments in renewables, chiefly with the help of China.⁶⁸ China's involvement in the African renewable energy sector did not originate overnight. The Chinese leadership's growing concern about the impact of climate change – partly fed by growing popular discontent about the level of air pollution – have pushed the country in the direction of an aggressive policy that promotes renewable energy.⁶⁹ Today, China is the world leader in domestic investment in renewable energy and associated low-emissions energy sectors, and

increasingly, the country has been investing in renewable energy projects abroad with the aim of expanding its renewable sector. Just last year, Chinese foreign investment in renewables grew by 60% to reach US\$ 32 billion, including eleven new overseas investment deals worth more than US\$ 1 billion each.⁷⁰

Around a third of new generation capacity additions in Sub-Saharan Africa in the period 2010-2015 were installed with Chinese companies acting as the main contractor,⁷¹ equal to the total contribution of the next four largest countries combined. More than half of these additions came from renewable sources, primarily hydro energy, where Chinese companies are dominant. At the same time, Chinese involvement in wind energy in South Africa and Ethiopia is increasing. With respect to solar energy, Chinese companies have hitherto primarily acted as PV suppliers, rather than construction contractors. Primary point of entry for Chinese companies in the Sub-Saharan African market has been South Africa, where Chinese PV manufacturer Jinko Solar has invested in a PV factory with an annual capacity of 120MW.⁷²

5.2 Solar energy potential and investments

Given the high number of sunshine hours on average in the Sub-Saharan region, the potential for solar energy is large (Figure 30).



FIGURE 30: ANNUAL SUNSHINE HOURS MAP OF THE WORLD

Indeed, in many countries its potential has already been acknowledged. Nevertheless, at this point solar energy's contribution to the region's energy mix is still negligible. However, should planned investments in solar energy materialize, it could make up 12% of the total energy mix by 2040.⁷³ At this point, solar energy is significant in countries such as Angola and Mali, in the latter of which it could make up 4% of the

energy mix by 2020.⁷⁴ South Africa is by far the most important country on the continent when it comes to solar energy production, accounting for 65% (1361 MW) "of the continent's cumulative installed solar PV capacity."⁷⁵

In many countries, solar energy is seen as a source with great potential, and projects aimed at fulfilling the potential have been launched. However, concrete results so far have remained modest. One example is Senegal, which will soon inaugurate a solar power plant north of Dakar that is to provide power to 180,000 local residents.⁷⁶ The investment comprises EUR 43 million, partly funded by the French Development Agency.⁷⁷ While relatively sizeable for a solar project, its impact on energy consumption will still be rather small overall. Other countries where projects are under way are Botswana⁷⁸ and Equatorial Guinea.⁷⁹ The continent's largest planned solar project, Desertec, ultimately failed due to costs and political issues.⁸⁰ Rather than through large-scale projects, solar appears more suitable for the decentralized provision of electricity, a distribution method particularly useful for rural areas.

5.3 Wind energy potential and investments

Overall, the potential for wind energy in Sub-Saharan Africa is rather minimal. This is chiefly due to issues of cost and relatively low wind speeds recorded in many parts of the continent. Exceptions to the rule are Angola, which aspires to build a windfarm in the southern desert⁸¹, Kenya, where a 310 MW strong wind park is being built at Lake Turkana, and Ethiopia, which currently has 171 installations across the country.⁸² Projects are also under way, or have been completed, in Nigeria, Senegal, and South Africa. The projects in South Africa should enable the country to add 8,400 MW to its energy mix by 2030.⁸³ Sudan is another country with significant potential. Here, the plan is to add 100MW to the grid by 2030.⁸⁴

The variability of electricity generated by wind power makes it a less reliable form of energy as a baseload. According to the IEA, "domestic markets are small and the power grids are not well developed, meaning that variable generation from wind would introduce additional challenges to an already unstable and intermittent system".⁸⁵ Also, the costs involved in building wind farms make it a less appealing source of energy for the region.

5.4 Biomass

Biomass is still the most important energy source for countries in Sub-Saharan Africa. Its use is mainly residential and for cooking purposes. At present, it makes up over 60% of total energy use.⁸⁶ In Africa as a whole, some 80% of the population is said to rely on biomass for daily purposes.⁸⁷ Unfortunately, among renewable energy sources, wood-based energy is considered dirty, inefficient and unsustainable. There is therefore a growing chorus for it to be replaced as quickly as possible by alternative energy sources.⁸⁸ On average, the use of biomass is higher in poorer countries. For instance, in Burundi, Rwanda, and the Central African Republic, the share of energy derived from biomass is 90% or higher.⁸⁹ Gabon, a country where oil consumption is significant at 29%, also relies on biomass for 62% for its primary energy supply.⁹⁰ Similar figures can be found in Gambia.⁹¹ Biomass is also a significant source of energy in large countries. In Ethiopia for instance, biomass makes up 92% of the domestic energy consumption, whilst hydropower is becoming increasingly important as well.⁹² In Kenya the percentage is 76%.⁹³

Biomass suffers from negative perceptions, being seen as outdated, and is therefore often presented as an "unsustainable source of energy, with its use potentially causing an energy crisis and even desertification."⁹⁴ This means that fossil fuels and nuclear energy can be more attractive alternatives.⁹⁵ However, others argue in favor of biomass use in Sub-Saharan Africa because of availability, familiarity and price, employment creation, energy security and diversity, climate change mitigation, potential for technological advancement and commercial investment, cleanliness and modernity.⁹⁶ Of course, whether these advantages can be reaped depends on how biomass is consumed. This makes it all the more important to make sure that biomass use becomes less onerous for people's health. The promotion of more efficient biomass cooking stoves can reduce the harmful health effects of pollution from indoor smoke. Nevertheless, 650 million people – more than one-third of an expanding population – will still cook with biomass in an inefficient and hazardous way in 2040.⁹⁷

If less harmful ways of consuming biomass can be developed, the potential for its future use is considerable. Bioenergy could sustainably contribute between a quarter and a third of global primary energy supply in 2050. It is the only renewable source that can replace fossil fuels in all energy markets – in the production of heat, electricity, and fuels for transport.⁹⁸ Projections suggest that the primary energy mix of sub-Saharan Africa will continue to be dominated by bioenergy. Demand is set to grow by 40%, to reach 490 Mtoe in 2040.⁹⁹ At the same time, the overall share of bioenergy in the primary energy mix is set to decline over time, from 61% in 2012 to 47% in 2040.¹⁰⁰

5.5 Local/decentralized energy generation

Sub-Saharan Africa is far from achieving universal energy access for its citizens. Energy poverty is widespread as a result. National grids often only serve urban areas, and fail to reach into the countryside. At present, only 18% of the rural population in Sub-Saharan Africa has access to electricity, compared to 55% in urban areas.¹⁰¹ In CAR only 3% of the population is plugged into the electricity grid.¹⁰² On average, electrification rates remain well below 50%. Only South Africa (85%) and Gabon (86%) achieve high figures. The overall electrification rate for Sub-Sahara Africa is 35%, with only 19% coverage in rural areas.¹⁰³ The total figure is set to increase to 71% by 2040,¹⁰⁴ with increases to 93% in urban and peri-urban areas and to 46% in rural areas.¹⁰⁵ However, significant efforts are needed to realize these percentages.

The provision of electricity to rural areas proceeds largely in a decentralized manner, with communities relying on off-grid mini-networks. Most of these networks run either on hydro or solar PV. An example thereof can be found in Angola, where the government plans to service 31 communities through mini-hydro installations¹⁰⁶, to be complemented by so-called solar villages.¹⁰⁷ Another example is CAR which is implementing a scheme for micro-hydroelectric power plants¹⁰⁸, and Ethiopia which, with the support of the World Bank, is rolling out off-grid solar lighting systems, being at the forefront of such initiatives in Africa.¹⁰⁹ It is estimated that in Sub-Saharan Africa, by 2040, solar photovoltaics, small hydropower or wind will power two-thirds of mini-grid and off-grid systems in rural areas.¹¹⁰

It is by no means a given that off-grid energy supply can be implemented successfully. For instance, in Nigeria, an initiative to install solar mini-grids largely failed as private investors could not be enticed to invest in the project given that electricity is seen as 'charity', i.e. that this should be provided for free by the government in the first place.¹¹¹ In Botswana, it was concluded that an extension of the existing national grid pays more dividends, and that off-grid electrification is only appropriate when used in remote villages.¹¹² In spite of the various initiatives across Sub-Sahara Africa, much of the rural population remains deprived of reliable access to electricity. Cost is a major factor, as the creation of off-grid systems, albeit attractive in some respects, continues to be more expensive than being plugged into a national grid.

5.6 Risks of making the wrong choices

The goal of increasing the use of renewable energy resources can be pursued for two main reasons. There are, first, environmental reasons and, second, the aspiration to create a more shock-resistant economy through diversification. The general trend among Sub-Saharan countries is to move away from non-renewable energy sources. This is done mainly in order to decrease the dependency on one particular resource and to reduce the vulnerability to price fluctuations in the oil market in particular.

Indeed, a number of countries in the region remain highly dependent on energy exports: oil is dominant in Angola, where it makes up 40% of all domestic energy consumption (2012) and 97% of export revenues (2012)¹¹³; Equatorial Guinea, where oil and gas make up 99% of all export earnings (2011); and Nigeria, where oil constitutes 90% of export earnings (2012).¹¹⁴ Conversely, some countries exhibit a high energy import dependency. One example is Botswana, where 44.2% of all energy sources are imported (2013).

Most countries in Sub-Saharan Africa have renewable energy promotion schemes in place, often using fiscal incentives or energy savings targets (e.g. using net metering).¹¹⁵ For some countries turning to renewable energy, this can and has entailed radical change in their economic landscape. Ethiopia for example became virtually energy independent through their investment in hydroelectricity and the country is transforming itself into an energy exporter. However, in this particular case the high dependency on hydro power makes Ethiopia vulnerable to climate conditions such as drought.¹¹⁶ This is an important reason why the government seeks to further diversify its energy supply and include other renewable energy sources, in particular geothermal energy.¹¹⁷ Total power sector investment averages around US\$ 46 billion per year, with just over half of it in transmission and distribution."¹¹⁸

Policies aimed at increasing the share of renewables in the energy mix of Sub-Saharan African countries, or to promote more environmentally-friendly energy sources can come under pressure from discoveries of fossil fuel fields. Various finds have been made in recent years off the coasts of Senegal, Ghana, Angola, and Mozambique, among others. However, low oil prices have curtailed exploration and development activity in oil and gas. What is more, continuing fiscal consolidation in oil-exporting countries is expected to result in further capital expenditure cuts.¹¹⁹ This means that the chances for these countries to successfully exploit offshore hydrocarbon reserves are hampered in the current low pricing environment. That said, this may change pending a rise in the price of oil.

In Senegal, Texas-based Kosmos Energy and UK-listed Cairn energy are said to want to push ahead to production while oil prices, and industry costs, are still low. However, many Senegalese fear that the country has a lot to lose if it were to go down the road of hydrocarbon exploration. With an economy set to grow at 6% this year underpinned by a strong agriculture sector, the country has been spared by some of the negative effects of the commodity slump that hit many other countries in Africa. With many of the critical decisions yet to be taken, such as what percentage of the revenues will be deposited in a fund for future generations, yet to be taken, there is skepticism about the extent to which Senegal is ready to handle a future revenue stream coming from oil and gas exports.¹²⁰

Despite the low pricing environment, a number of low-income, non-oil commodity exporters are expected to continue to invest heavily in energy and transport infrastructure in a bid to improve the operational environment for growth.¹²¹ One example is Gabon, which has initiated an energy diversification strategy as oil prices started to drop. It attempts to ramp up hydroelectric power in particular.¹²² South Africa, the only country in the region that relies heavily on coal, has set itself a target of having approximately two-thirds of all domestic energy use generated through renewables by 2030.¹²³ A significant continent-wide initiative is the Climate Vulnerable Forum, which includes countries such as the Democratic Republic of Congo, Gambia, Ghana, Kenya, Senegal, and Sudan and aims to phase out the use of oil and coal as part of the energy mix and to rely exclusively on renewables. A pledge to that effect was affirmed in Marrakech in November 2016.¹²⁴ There are other encouraging signs. For instance, investment in environmentally-responsible energy such as by the Investor Platform for Climate Action, a group which controls assets worth US\$ 25 trillion, can make a difference in shifting funding away from coal, for instance by encouraging divestments away from coal assets and encouraging companies to invest in sustainable infrastructure.¹²⁵ Also, the exportation of renewable energy is taking off through the African Powerpool collaboratives (West-African, South African, and East African Power Pools), which seek to integrate regional electricity markets and create common power arids.¹²⁶

An absolute prerequisite for countries to enhance their economic growth is the access of people and companies to the electricity grid.¹²⁷ What still remains a problem however is investor risk, in part because many countries lack sufficient institutional policy support for green measures.¹²⁸ Partly as a consequence, the private financing of infrastructure that is high-carbon, not climate-resilient, or generally unsustainable still significantly outweighs private investment flows into sustainable infrastructure.¹²⁹ Moreover, many Sub-Saharan countries still face difficulties regarding the upkeep (e.g., Nigeria), extraction and optimization of capacity. Nigeria is an extreme example given that it has a 75% loss of energy before it gets the supplies to the consumer.

Conclusions

In conclusion, it is clear that the abundance of biomass and its continuing potential for local use will endure for as long as national grids are not extended into rural areas. This

means that the major challenge for Sub-Saharan African countries in the future will be to reduce the hazardous health effects resulting from the use of biomass. On top of providing more efficient biomass cooking stoves, a more sustainable solution should be found in the provision of decentralized electricity generation via off-grid mininetworks that function on the basis of either hydro energy or solar PV.

6 CONCLUSIONS AND RECOMMENDATIONS

6 CONCLUSIONS AND RECOMMENDATIONS

This study has sought to investigate the geopolitical implications of climate mitigation policies that aim to bring about a transition away from fossil fuels and towards a low-carbon economy. The results of our quantitative analysis indicate that in all future scenarios where climate mitigation policies are successful, a relative increase in instability in the countries investigated – Algeria, Egypt, Kazakhstan, Libya, Qatar, Russia and Saudi Arabia – could be recorded. In other words, the second order effects of existing climate mitigation policies can be considerable and, therefore, should be taken into account by policy makers.

Conclusions

Declining imports of fossil fuels in Europe will result in a reduction of the mutual dependencies that exist between Europe and the hydrocarbon exporting countries in our immediate neighborhood. One consequence thereof, in the absence of farreaching reforms, is a higher risk of instability in the latter countries. In the longer term, this reduced economic dependency will also reduce the influence that the Netherlands, Europe and the Western world more generally have on these countries. The logic is simple: if the importance of the European market decreases relative to other markets, it means that the ability of Europe to exert diplomatic pressure on large hydrocarbon states in the Middle East and in the region belonging to the former Soviet Union recedes. Other players, such as China and India may fill the void. To provide an example, the 'threat' to diversify our supply base and reduce our dependency on Russia therefore becomes a less potent tool in our foreign policy make-up in the case of future disagreements with the Kremlin.

What is worrisome is that if – over time – Europe proved to be the only region worldwide imposing a legally enforceable CO_2 emissions cap while and other countries continued to use consequentially cheaper fossil fuels, this would hamper the competitive position of Europe as a whole. It is, therefore, of the utmost importance
that in future climate change negotiations, talks of a carbon price or a legally enforceable CO₂ emissions cap aim at the global level.

A lack of attention for the second order effects of energy transition in the form of heightened instability, in particular in the Middle-East and North Africa, runs the risk of creating adverse geopolitical consequences in the form state failure and collapse comparable to the 2011 Arab uprisings. If these consequences materialized, they would further undermine the possible willingness on the part of countries richly endowed in hydrocarbons to push for a more ambitious climate policy.

Critical factors

A number of factors emerged from the study as 'critical' in terms of their influence in determining as to whether a hydrocarbon exporting country is prone to the destabilizing effects of climate mitigation policies.

The first critical factor are *energy prices*. If in the future the international community decides to install a legally enforceable cap on carbon emissions this may cause a new, much deeper decline of energy prices. Interesting in this regard is the observation that the coal market can be seen as being 'ahead of the curve' in the sense that developments in the coal market show a comparable cycle to those in the oil market. In other words, if prices on the coal market plummet, it is reasonable to assume that the oil market will follow.

Second, when it comes to the *stability* of rentier states, it should be considered to develop a 'score card' for the relative vulnerability of hydrocarbon exporting states. Specific indicators that mark a heightened vulnerability are: a high share of resource rents in GDP in combination with limited financial reserves; a high national debt as percentage of GDP or a rapid increase in a state's national debt; a young and/or relatively fast growing population (Middle East and North Africa), a relatively high share of youth unemployment and uncertainty about the continuation of the existing political leadership; a combination of high subsidies and a high domestic energy demand; and the position in international financial markets (risk-aggravating factors are a low credit rating and difficulties to borrow in international markets). It should be stressed that having a good position in international financial markets and ample financial reserves, as in the case of Saudi Arabia, may provide relief in the short to medium term, yet it by no means makes a country immune to the destabilizing effects of climate mitigation policies and energy transition if significant shortcomings are recorded in regards to the other critical factors.

Recommendations

The analysis conducted in this report leads us to a set of policy recommendation and policy priorities with the aim of promoting desirable scenarios and limiting the negative security effects that result from undesirable scenarios.

In the context of energy transition it is important to pre-empt the arrival of a legally enforceable carbon cap. For the Dutch economy, this can be done by promoting energy efficiency measures and the greening of the domestic economy in order to increase its competitiveness. The ambition of the Dutch government to greatly reduce our dependence on natural gas until 2050 can thus be viewed as a welcome initiative in this regard.

At the same time, we should anticipate that the incentive for countries to again resort to fossil fuels will increase when prices will decrease as a result of a legal cap on CO₂ emissions in some parts of the world. Countries not covered by the cap may be tempted to increase their use of fossil fuels and countries under the cap system may be incentivized to withdraw. Internationally, this will require lobbying at the highest level to motivate countries to take part in a legal carbon cap system and continued investment in renewable energy with the aim of cutting costs. For this purpose, it is of major importance to create *prospects of gain*: both the Netherlands and the EU should be able to demonstrate clearly the economic benefits of decoupling and of a transition towards renewable energy.

A different but related challenge manifests itself in developing countries that face a choice between investing in renewable energy networks or to rely on fossil fuels for their economic growth. A major issue in this context is the need to reduce the high use of biomass in Sub-Saharan Africa, owing to its hazardous health effects and the problems associated with indirect land use change. In the long term it is crucial that a lower use of biomass does not result in a higher reliance on fossil fuel resources. This means that, particularly when viewed through the lens of ensuring coherence in Dutch international policies, the government will have to determine how to deal with public (financial) bilateral contributions to the exploration of new hydrocarbon finds in developing countries. In this context, the 'Energy Agenda' (Dutch: Energie-agenda) of December 2016 states that in order to safeguard coherence in international policies, the government how to assess its public bilateral contributions to wards the exploration of new reserves of fossil fuels in light of the remaining / available emission space. ¹³⁰ This means that the Dutch government should carefully balance its contributions to the development of recently discovered hydrocarbon

reserves against those in the sphere of renewable energy, the circular economy and clean-tech. Not only is this defensible from an environmental point of view, but it could also benefit the competitive position of Dutch companies active in these industries in the long term.

In the absence of wide-reaching reforms, many of the countries studied will face scenarios in which they will struggle to deal with domestic unrest. The biggest risk thereof will manifest itself in the Middle East and North Africa. A silver lining considering energy transition is that, as the name of the phenomenon implies, this is not something that will happen overnight. Rather, it is expected that the completion of a transition away from fossil fuels will take several decades. This means that there is still time for countries heavily dependent on the export of fossil fuels to take adequate measures to reduce their vulnerability.

This however requires a fundamental reassessment of the nature of the relationship that we have with this part of the world. The crux of the matter is that whereas western nations have an interest in the Gulf countries remaining stable, the Gulf countries themselves have a key interest in reducing their vulnerability with an eye on future regime survival. When reforming the domestic economies of these countries, the focus should be on addressing the factors labelled as critical in this chapter.

Switching from fossil fuel-based power generation to one based on renewable energy can, on the one hand, mean a significant cost reduction by eliminating wasteful energy subsidies, whilst reducing the burning of otherwise valuable export resources on the other. In the bilateral ties and sectoral dialogues between the Netherlands and countries in the MENA region – the Gulf countries in particular – it is important to stress that doing so represents a significant cost cutting measure at a time when export revenues are under pressure, and would also allow for the diversification of the economy through the buildup of a domestic renewable energy sector that can compete internationally. For the Netherlands this represents a business opportunity to showcase technological advances in the areas of renewable energy, energy efficiency and clean-tech.

In convincing countries that it is in their interest to do so it is useful to take note of the Chinese experience. Gulf countries with significant sovereign wealth funds, such as Saudi Arabia and Qatar, can learn a thing or two from how China built up a strong domestic renewable energy sector, followed by a sizeable overseas expansion strategy in frontier markets. If they succeed in building up a domestic renewable energy sector,

it would be beneficial as a follow-on step to invest in the energy transition in hydrocarbon exporting countries that are financially less well-endowed and in relative proximity to the European mainland, such as Algeria and Egypt, so that the export of electricity becomes feasible. This would help rentier states around Europe to lessen their dependence on commodity sales to Europe. More importantly, it would dampen the need for domestic energy subsidies and reduce the negative effects that climate mitigation policies will have on these countries. For Saudi Arabia and Qatar, it would result in an ability to collect a return on their investments and to generate additional export revenues allowing these countries to *hedge* against lower oil prices and diminished income from hydrocarbon exports in the future. For Europe, this would mean that – provided the required infrastructure is financed – it can tap into an additional source of renewable energy that is located in relative proximity, whilst the risk of large-scale instability in our immediate neighborhood is reduced.

BIBLIOGRAPHY

BIBLIOGRAPHY

- Abdeladim, Kamel, Salim Bouchakour, Amar Hadj Arab, Saïd Ould Amrouche, Farida Cherfa, Bilel Taghezouit, and Karim Kerkouche. 'RENEWABLE ENERGIES IN ALGERIA: CURRENT SITUATION AND PERSPECTIVES.' In *ResearchGate*, 2014. doi:10.13140/2.1.1600.4808.
- African Development Bank Group. "Renewable Energy in Africa- Mali Country Profile." Renewable Energy in Africa. African Development Bank Group, 2015. http://www.afdb.org/ fileadmin/uploads/afdb/Documents/Generic-Documents/Profil_ER_Mal_Web_light.pdf.
- "Africa's Largest Microgrid Takes Shape In Equatorial Guinea | Engerati Generation." *Engerati*, December 16, 2014. https://www.engerati.com/article/africa%E2%80%99s-largestmicrogrid-takes-shape-equatorial-guinea.
- Alexovich, Ariel. "Paris Climate Agreement to Enter into Force on 4 November." *United Nations Sustainable Development*, October 5, 2016. http://www.un.org/sustainabledevelopment/blog/2016/10/paris-climate-agreement-to-enter-into-force-on-4-november/.
- "Angola- U.S. Energy Information Administration (EIA)." *Marcon*. Accessed October 17, 2016. http://www.marcon.com/marcon2c.cfm?SectionListsID=30&PageID=2262.
- Angolan Ministry of Energy and Water. "RURAL ELECTRIFICATION." *Republica de Angola Ministerio Da Energia E Aguas*, 2016. http://www.angolaenergia2025.com/en/conteudo/rural-electrification.
- Auping, Willem L., Erik Pruyt, Sijbren De Jong, and Jan H. Kwakkel. "Simulating the Impact of Climate Mitigation Policies on Social Unrest in Rentier States." Delft, Netherlands: System Dynamics Society, 2016.

- ———. "The Geopolitical Impact of the Shale Revolution: Exploring Consequences on Energy Prices and Rentier States." Energy Policy. Netherlands, 2016.
- Bauen, Ausilio, Göran Berndes, Martin Junginger, Marc Londo, and François Vuille. 'Bioenergy-A Sustainable and Reliable Energy Source. A Review of Status and Prospects.' IEA, June 2009. http://www.seai.ie/Renewables/Bioenergy/Bioenergy_%E2%80%93_a_Sustainable_ and_Reliable_Energy_Source_MAIN_REPORT.pdf.
- Bojanowski, Axel. 'Kampf Gegen Klimawandel: 45 Länder Wollen Komplett Auf Kohle, Öl Und Gas Verzichten.' *Spiegel Online*, November 18, 2016. http://www.spiegel.de/wissenschaft/ natur/klimakonferenz-in-marrakesch-45-laender-verzichten-komplett-auf-kohle-oel-undgas-a-1121838.html.
- "Botswana 10th National Development Plan 2009-2016," 2009. http://www. nationalplanningcycles.org/sites/default/files/country_docs/Botswana/ndp_botswana.pdf.
- Bryant, Benjamin P, and Robert J. Lempert. "Thinking inside the Box: A Participatory, Computer-Assisted Approach to Scenario Discovery." *Technological Forecasting & Social Change* 77 (2010): 34–49. doi:10.1016/j.techfore.2009.08.002.
- Buckley, Tim, and Simon Nicholas. "China's Global Renewable Expansion. How the Worl's Second-Biggest National Economy Is Positioned to Lead the World in Clean-Power Investment." Institute for Energy Economics and Financial Analysis, January 2017.
- Butler, Nick. "China's Energy Strategy: Power and Independence." *Financial Times*, January 16, 2017. http://blogs.ft.com/nick-butler/2017/01/16/chinas-energy-strategy-power-and-independence/.
- Calderbank, Selwa. "Desertec Abandons Sahara Solar Power Export Dream." *EurActiv*, May 31, 2013. http://www.euractiv.com/section/trade-society/news/desertec-abandons-sahara-solar-power-export-dream/.
- CAR Ministry of Mines Energy and Hydraulics. "Expression of Interest by the Central African Republic to CIF/SREP Program." World Bank Group, 2014. 17-10-2016. https://www-cif. climateinvestmentfunds.org/sites/default/files/meeting-documents/car_eoi_0.pdf.

- Castellano, Antonio, Adam Kendall, Mikhail Nikomarov, and Tarryn Swemmer. "Brighter Africa." *McKinsey*, February 2015. http://www.mckinsey.com/industries/electric-power-and-naturalgas/our-insights/powering-africa.
- Collier, Paul, and Anke Hoeffler. "Greed and Grievance in Civil Wars." *Oxford Economic Papers* 56 (2004): 563–95.
- "CVF Participating Countries." *Climate Vulnerable Forum*. Accessed November 18, 2016. http:// www.thecvf.org/web/climate-vulnerable-forum/cvf-participating-countries/.
- Deutsche Gesellschaft für Internationale Zusammenarbeit. "The Nigerian Energy Sector- An Overview with a Special Emphasis on Renewable Energy, Energy Efficiency and Rural Electrification." Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ), June 2015. https://www.giz.de/en/downloads/giz2015-en-nigerian-energy-sector.pdf.
- Diallo, Amadou Oury. "Mali: un projet de centrale solaire de 50MW a Kita." *Jeune Afrique*, November 5, 2015. https://www.akuoenergy.com/fileadmin/media/pdf/press/2015-11-05-5959@WWW_JEUNEAFRIQUE_COM.pdf.
- "Distance to Frontier Doing Business World Bank Group." *The World Bank*, 2016. http://www. doingbusiness.org/data/distance-to-frontier.
- "Drought: Hydropower's Achilles Heel." *International Rivers*, May 16, 2016. https://www. internationalrivers.org/blogs/433/drought-hydropower-s-achilles-heel.
- "Eastern Africa Power Pool." *Eastern Africa Power Pool*. Accessed December 23, 2016. http:// eappool.org/.
- "Energieagenda. Naar Een CO'-Arme Energievoorziening." Den Haag: Ministerie van Economische Zaken, December 2016.
- "Energierapport Transitie Naar Duurzaam." Ministerie van Economische Zaken, January 2016.
- Energy Transition Advisors. "Carbon Tracker Initiative." Energy Access. ETA, November 2014. https://www.carbontracker.org/wp-content/uploads/2014/11/Coal-Energy-Access-111014final.pdf.

- FAO, IFAD, IMF, OECD, UNCTAD, WFP, World Bank, WTO, IFPRI, and UN HLTF. "Price Volatility in Food and Agricultural Markets: Policy Responses," June 2, 2011. http://www.oecd.org/ trade/agricultural-trade/48152638.pdf.
- Federal Democratic Republic of Ethiopia Ministry of Water and Energy. "Scaling-up Renewable Energy Program Ethiopia Investment Plan (Draft Final)." OECD, January 2012. https://www. oecd.org/env/cc/TADELE_FDRE%20Ethiopia%20Scaling%20-%20Up%20Renewable%20 Energy%20Program%202012.pdf.
- Fick, Maggie. "Oil and Gas Discoveries Fuel Senegalese Fears." *Financial Times*, December 27, 2016. https://www.ft.com/content/065ba414-bbce-11e6-8b45-b8b81dd5d080.
- Forrester, J.W. Industrial Dynamics. Cambridge, Massachusetts: MIT Press, 1961.
- "Global Trends in Renewable Energy Investment 2016." Frankfurt: Frankfurt School FS-UNEP Collaborating Centre for Climate & Sustainable Energy Finance, 2016. http://fs-unepcentre.org/publications/global-trends-renewable-energy-investment-2016.
- "Guidelines to Renewable Energies." People's Democratic Republic of Algeria, Ministry of Energy and Mines, n.d.
- International Energy Agency. "Africa Energy Outlook." *International Energy Agency*, 2014. https://www.iea.org/publications/freepublications/publication/weo-2014-special-reportafrica-energy-outlook.html.
 - - —. "IEA Report- Senegal." International Energy Agency, 2014. https://www.iea.org/ statistics/statisticssearch/report/?country=Senegal&product=electricityandheat.
 - —. "World Energy Outlook Energy Access Database." Database. *International Energy Agency*, 2016. http://www.worldenergyoutlook.org/resources/energydevelopment/energyaccessdatabase/.

- "International Energy Agency Statistics." *International Energy Agency (IEA)*, 2016. https://www.iea.org/statistics/.
- International Renewable Energy Agency. "Africa 2030: Roadmap for a Renewable Energy Future," 2015. http://www.irena.org/menu/index. aspx?mnu=Subcat&PriMenuID=36&CatID=141&SubcatID=641.
- —. "Renewable Energy Country Profiles Africa." International Renewable Energy Agency, 2011. http://www.irena.org/DocumentDownloads/Publications/CountryProfiles_Africa_ WEB.pdf.
- . "Solar PV in Africa: Costs and Markets." Accessed October 6, 2016. http://www.irena. org/menu/index.aspx?mnu=Subcat&PriMenuID=36&CatID=141&SubcatID=2744.
- Jaeger, Joel, Paul Joffe, and Ranping Song. "China Is Leaving the U.S. Behind on Clean Energy Investment | World Resources Institute," January 6, 2017. http://www.wri.org/blog/2017/01/ china-leaving-us-behind-clean-energy-investment.
- Jong, Sijbren de, Willem Lucas Auping, and Joris Govers. "The Geopolitics of Shale Gas. The Implications of the US' Shale Gas Revolution on Instrastate Stability within Traditional Oil and Natural Gas Exporting Countries in The EU Neighborhood." Strategy & Change. The Hague: The Hague Centre for Strategic Studies, 2014.
- Kenning, Tom. "Kenya to Construct 55MW Solar Plant with Chinese Government Funding." *PV-Tech*, April 4, 2016. http://www.pv-tech.org/news/kenya-to-construct-55mw-solar-plantwith-chinese-government-funding.
- Kerr, Simon. "Saudi Arabia Seeks \$30bn-\$50bn Solar and Wind Energy Investment." Financial Times, January 16, 2017. https://www.ft.com/content/d370829e-dbfe-11e6-86acf253db7791c6.
- KPMG Africa Infrastructure & Major Projects Gropu. "Sub-Saharan Africa Power Outlook 2014." KPMG, 2014. http://www.kpmg.com/ZA/en/IssuesAndInsights/ArticlesPublications/ General-Industries-Publications/Documents/2014%20Sub-Saharan%20Africa%20 Power%20Outlook.pdf.
- "Lake Turkana Wind Power | Project Overview," 2017. http://ltwp.co.ke/.

- Laleix, Gaëlle. 'La plus grande centrale solaire d'Afrique de l'Ouest a trouvé ses financements.' *RFI*, May 17, 2016. http://www.rfi.fr/emission/20160517-plus-grande-centrale-solaireafrique-ouest-trouve-financements.
- Lempert, Robert J., David G. Groves, Steven W. Popper, and Steven C. Bankes. "A General, Analytic Method for Generating Robust Strategies and Narrative Scenarios." *Management Science* 52, no. 4 (2006): 514–28. doi:10.1287/mnsc.1050.0472.
- Lempert, Robert J., Steven W. Popper, and Steven C. Bankes. "Shaping the next One Hundred Years : New Methods for Quantitative, Long-Term Policy Analysis." Santa Monica: RAND, 2003. http://www.rand.org/pubs/monograph_reports/2007/MR1626.pdf.
- Mahdavy, H. "The Pattern and Problems of Economic Development in Rentier States: The Case of Iran." In *Studies in the Economic History of the Middle East.*, edited by M. A. Cook. Oxford: UK: Oxford University Press., 1970.
- "Medium-Term Renewable Energy Market Report 2016." Paris: International Energy Agency (IEA), 2016. http://www.iea.org/bookshop/734-Medium-Term_Renewable_Energy_Market_ Report_2016.
- Mitchell, Donald. "A Note on Rising Food Prices." Policy research working paper. The World Bank, July 2008. http://www-wds.worldbank.org/servlet/WDSContentServer/WDSP/IB/200 8/07/28/000020439_20080728103002/Rendered/PDF/WP4682.pdf.
- "Nationale Energieverkenning 2016." ECN, Planbureau voor de Leefomgeving (PBL), Centraal Bureau voor de Statistiek (CBS)en de Rijksdienst voor Ondernemend Nederland, 2016.
- "Natural Gas Rents (% of GDP) | Data." *World Bank*. Accessed December 22, 2016. http://data. worldbank.org/indicator/NY.GDP.NGAS.RT.ZS.
- Ndiaye, Amadou. 'Le Sénégal se dote de sa première centrale solaire à Bokhol.' *Le Monde*. March 23, 2016. http://www.lemonde.fr/afrique/article/2016/03/23/le-senegal-se-dote-desa-premiere-centrale-solaire-a-bokhol_4888448_3212.html.
- "Nederland Moet Uitstoot Broeikasgas CO2 Flink Terugbrengen." NOS, June 24, 2015. http:// nos.nl/artikel/2043165-nederland-moet-uitstoot-broeikasgas-co2-flink-terugbrengen.html.

- OECD. "Biodiesel Production, OECD Countries, 1990-2011." *OECD Publishing*, 2013. http:// www.keepeek.com/Digital-Asset-Management/oecd/agriculture-and-food/oecdcompendium-of-agri-environmental-indicators/biodiesel-production-oecd-countries-1990-2011_9789264186217-table33-en#page1.
- "Oil Rents (% of GDP) | Data | Table." *World Bank*, 2012. http://data.worldbank.org/indicator/ NY.GDP.PETR.RT.ZS.
- Okafor, Chineme. "Nigeria's Leap Towards Solar Energy." *This Day (Lagos)*, July 19, 2016. http://allafrica.com/stories/201607190987.html.
- Oreskes, Naomi, Kristin Shrader-Frechette, and Kenneth Belitz. "Verification, Validation, and Confirmation of Numerical Models in the Earth Sciences." *Science* 263, no. 5147 (1994): 641–46. doi:10.1126/science.263.5147.641.
- Overmars, Koen, Robert Edwards, Monica Padella, Anne Gerdien Prins, and Luisa Marelli. "Estimate of Indirect Land Use Change from Biofuels Based on Historical Data." JRC Science and Policy Briefs. European Commission, n.d.
- Owen, Matthew, Robert van der Plas, and Steve Sepp. "Can There Be Energy Policy in Sub-Saharan Africa without Biomass?" *Energy for Sustainable Development*, Special Issue on Charcoal, 17, no. 2 (April 2013): 146–52. doi:10.1016/j.esd.2012.10.005.
- Oxford Business Group. "Gabon Aspires to Emerging Market Status." Oxford Business Group, August 18, 2015. https://www.oxfordbusinessgroup.com/analysis/reviewing-plan-progresstowards-goal-reaching-emerging-market-status.
- "Paris Agreement." United Nations, 2015.
- Philip N. Johnson-Laird. *Mental Models: Towards a Cognitive Science of Language, Inference, and Consciousness*. Cambridge, MA: Harvard University Press, 1983.
- "Polity IV Project: Home Page." *Center for Systemic Peace*, June 10, 2013. http://www.systemicpeace.org/polity/polity4.htm.
- "Polity Project." *The Center for Systemic Peace*, 2016. http://www.systemicpeace.org/ polityproject.html.

- "Power Africa in Ethiopia." USAID, September 28, 2016. https://www.usaid.gov/powerafrica/ ethiopia.
- "Private Participation." Angola Energy 2025- Angola Power Sector Long Term Vision. Accessed October 26, 2016. http://www.angolaenergia2025.com/en/conteudo/private-participation.
- "Pros and Cons of Hydroelectric Power." *Conserve Energy Future*, February 17, 2015. http:// www.conserve-energy-future.com/pros-and-cons-of-hydroelectric-power.php.
- Rao, P Govinda. "Assessment of Solar Energy Potential in Qatar." presented at the Solar Qatar Summit, November 2013.
- Renewable Energy Policy Network for the 21st Century. "Renewables 2016 Global Status Report." Renewable Energy Policy Network for the 21st Century, June 2016. http://www. ren21.net/wp-content/uploads/2016/06/GSR_2016_Full_Report.pdf.
- "Renewable Energy Snapshot: Azerbaijan." United Nations Development Programme (UNDP), 2013.
- "Renewable Energy Snapshot: Kazakhstan." United Nations Development Programme (UNDP), 2013.
- "Renewable Energy: The Case of Botswana." Text/HTML. World Bank. Accessed September 29, 2016. http://www.worldbank.org/en/news/feature/2014/08/11/renewable-energy-the-case-of-botswana.
- Republic of Angola Ministry of Energy and Water. "The Renewable Energies in Angola Current Picture and Perspectives." n.d. http://www.energyafrica.de/fileadmin/user_upload/Energy-Africa__12/Presentation_Ministry%20of%20Energy%20and%20Water_The%20 renewable%20energies%20in%20Angola.pdf.
- Saibi, Hakim. "Geothermal Resources in Algeria." *Renewable and Sustainable Energy Reviews* 13, no. 9 (December 2009): 2544–52. doi:10.1016/j.rser.2009.06.019.
- "Saudi Arabia Foreign Exchange Reserves | 2010-2016 | Data | Chart." *Trading Economics*, 2016. http://www.tradingeconomics.com/saudi-arabia/foreign-exchange-reserves.

- "Saudi Arabia's Post-Oil Future." *The Economist*, April 2016. http://www.economist.com/news/ middle-east-and-africa/21697673-bold-promises-bold-young-prince-they-will-be-hard-keepsaudi-arabias.
- Sommer, Martin, Greg Auclair, Armand Fouejieu, Inutu Lukonga, Saad Quayyum, Amir Sadeghi, Shbaikat Gazi, Andrew Tiffin, Juan Trevino, and Bruno Versailles. "Learning to Live with Cheaper Oil - Policy Adjustment in Oil-Exporting Countries of the Middle East and Central Asia." International Monetary Fund Middle East and Central Asia Department, 2016. http:// www.imf.org/external/pubs/ft/dp/2016/mcd1603.pdf.
- South African Government. "Energy." *South African Government*. Accessed October 25, 2016. http://www.gov.za/about-SA/energy.

"Southern African Power Pool." Accessed December 23, 2016. http://www.sapp.co.zw/.

- Stecher, K., A. Brosowski, and D. Thrän. 'Biomass Potential in Africa.' IRENA and DBFZ, 2013. http://www.irena.org/menu/index. aspx?mnu=Subcat&PriMenuID=36&CatID=141&SubcatID=349.
- Sterman, John D. *Business Dynamics: Systems Thinking and Modeling for a Complex World.* Boston: McGraw-Hill, 2000.
- "The Agreement on Energy for Sustainable Growth: A Policy in Practice." The Hague: Sociaal-Economische Raad (Social and Economic Council), 2015.
- The Global Commission on Economy and Climate. "The Sustainable Infrastructure Imperative: Financing for Better Growth and Development. The 2016 New Climate Change Economy Report." The New Climate Economy. Washington/London: The Global Commission on the Economy and Climate, 2016. http://newclimateeconomy.report/2016/wp-content/uploads/ sites/4/2014/08/NCE_2016Report.pdf.
- "The Prodigal Sun: Renewables Have Huge Potential for Domestic Use, and Possibly Export. Algeria | Energy." *Oxford Business Group*, December 8, 2013. http://www. oxfordbusinessgroup.com/analysis/prodigal-sun-renewables-have-huge-potential-domesticuse-and-possibly-export.
- The World Bank. "Unemployment, Youth Total (% of Total Labor Force Ages 15-24) (Modeled ILO Estimate)," 2016. http://data.worldbank.org/indicator/SL.UEM.1524.ZS.

- "The World Factbook Central Intelligence Agency Ethiopia," January 12, 2017. https://www. cia.gov/library/publications/resources/the-world-factbook/geos/et.html.
- "The World Factbook CAR Central Intelligence Agency." Accessed October 5, 2016. https:// www.cia.gov/library/publications/the-world-factbook/geos/ct.html.
- Tlili, Iskander. "Renewable Energy in Saudi Arabia: Current Status and Future Potentials." Environment, Development and Sustainability: A Multidisciplinary Approach to the Theory and Practice of Sustainable Development 17, no. 4 (2015): 859–86.
- UNFCCC. "Kyoto Protocol to the United Nations Framework Convention on Climate Change." United Nations, 1998. http://unfccc.int/resource/docs/convkp/kpeng.pdf.
- ———. "United Nations Framework Convention on Climate Change." United Nations, 1992. https://unfccc.int/files/essential_background/background_publications_htmlpdf/application/ pdf/conveng.pdf.
- "U.S. Energy Information Administration (EIA) Country Reports," 2016. http://www.eia.gov/beta/ international/.
- Verlaan, Jos. "Kabinet Vecht Vonnis Aan in Klimaatzaak Urgenda." NRC Handelsblad, September 24, 2015. https://www.nrc.nl/nieuws/2015/09/24/het-kabinet-vecht-het-vonnis-aan-in-deklimaatzaak-a1412908.

"Vision 2030." Kingdom of Saudi Arabia, 2016.

- Vuuren, D. P. v., Elzen, M. G. J. d., Berk, M. M., Lucas, P., Eickhout, B., Eerens, H., and Oostenrijk, R. "Regional Costs and Benefits of Alternative Post-Kyoto Climate Regimes: Comparison of Variants of the Multi-Stage and Per Capita.," 2003. http://www.rivm.nl/en/Documents_and_ publications/Scientific/Reports/2003/november/Regional_costs_and_benefits_of_alternative_ post_Kyoto_climate_regimes_Comparison_of_variants_of_the_Multi_stage_and_Per_Capita_Convergence_regimes?sp=cml2bXE9ZmFsc2U7c2VhcmNoYmFzZT01OTM3MDtyaXZtcT-1mYWxzZTs=&pagenr=5938.
- World Bank. "Off-Grid Solar Lighting Up Ethiopia." *World Bank*, August 15, 2016. http://www. worldbank.org/en/news/feature/2016/08/15/off-grid-solar-lighting-up-ethiopia.

- World Bank Group. "Sub-Saharan Africa." Global Economic Prospects. World Bank Group, January 2016. https://www.worldbank.org/content/dam/Worldbank/GEP/GEP2016a/Global-Economic-Prospects-January-2016-Sub-Saharan-Africa-analysis.pdf.
- "World Energy Outlook 2016." Paris: International Energy Agency (IEA), 2016. http://www.worldenergyoutlook.org/publications/weo-2016/.

ENDNOTES

ENDNOTES

- 1. "Paris Agreement" (United Nations, 2015).
- Ariel Alexovich, "Paris Climate Agreement to Enter into Force on 4 November," United Nations Sustainable Development, October 5, 2016, http://www.un.org/sustainabledevelopment/blog/2016/10/paris-climateagreement-to-enter-into-force-on-4-november/.
- "The Agreement on Energy for Sustainable Growth: A Policy in Practice" (The Hague: Sociaal-Economische Raad (Social and Economic Council), 2015), 1.
- 4. "Energierapport Transitie Naar Duurzaam" (Ministerie van Economische Zaken, January 2016), 5.
- "Nationale Energieverkenning 2016" (ECN, Planbureau voor de Leefomgeving (PBL), Centraal Bureau voor de Statistiek (CBS)en de Rijksdienst voor Ondernemend Nederland, 2016), 16.
- 6. "Energieagenda. Naar Een CO₂-Arme Energievoorziening," 5.
- "Nederland Moet Uitstoot Broeikasgas CO2 Flink Terugbrengen," NOS, June 24, 2015, http://nos.nl/ artikel/2043165-nederland-moet-uitstoot-broeikasgas-co2-flink-terugbrengen.html.
- Jos Verlaan, "Kabinet Vecht Vonnis Aan in Klimaatzaak Urgenda," NRC Handelsblad, September 24, 2015, https:// www.nrc.nl/nieuws/2015/09/24/het-kabinet-vecht-het-vonnis-aan-in-de-klimaatzaak-a1412908.
- H. Mahdavy, "The Pattern and Problems of Economic Development in Rentier States: The Case of Iran.," in *Studies in the Economic History of the Middle East.*, ed. M. A. Cook (Oxford: UK: Oxford University Press., 1970).
- UNFCCC, "United Nations Framework Convention on Climate Change." (United Nations, 1992), art. 4.8, https:// unfccc.int/files/essential_background/background_publications_htmlpdf/application/pdf/conveng.pdf; UNFCCC, "Kyoto Protocol to the United Nations Framework Convention on Climate Change" (United Nations, 1998), http:// unfccc.int/resource/docs/convkp/kpeng.pdf.
- 11. Vuuren, D. P. v. et al., "Regional Costs and Benefits of Alternative Post-Kyoto Climate Regimes: Comparison of Variants of the Multi-Stage and Per Capita.," 2003, http://www.rivm.nl/en/Documents_and_publications/ Scientific/Reports/2003/november/Regional_costs_and_benefits_of_alternative_post_Kyoto_ climate_regimes_Comparison_of_variants_of_the_Multi_stage_and_Per_Capita_Convergence_ regimes?sp=cml2bXE9ZmFsc2U7c2VhcmNoYmFzZT010TM3MDtyaXZtcT1mYWxzZTs=&pagenr=5938.
- "International Energy Agency Statistics," International Energy Agency (IEA), 2016, https://www.iea.org/ statistics/.

- In 2013, 68% of population in Africa and 51% in developing Asia used traditional biomass for cooking. See Renewable Energy Policy Network for the 21st Century, "Renewables 2016 Global Status Report" (Renewable Energy Policy Network for the 21st Century, June 2016), http://www.ren21.net/wp-content/uploads/2016/06/ GSR_2016_Full_Report.pdf.
- "Medium-Term Renewable Energy Market Report 2016" (Paris: International Energy Agency (IEA), 2016), http:// www.iea.org/bookshop/734-Medium-Term_Renewable_Energy_Market_Report_2016.
- 15. Ibid., 7.
- "Global Trends in Renewable Energy Investment 2016" (Frankfurt: Frankfurt School FS-UNEP Collaborating Centre for Climate & Sustainable Energy Finance, 2016), http://fs-unep-centre.org/publications/global-trends-renewableenergy-investment-2016.
- "World Energy Outlook 2016" (Paris: International Energy Agency (IEA), 2016), http://www.worldenergyoutlook. org/publications/weo-2016/.
- 18. Ibid.
- Benjamin P. Bryant and Robert J. Lempert, "Thinking inside the Box: A Participatory, Computer-Assisted Approach to Scenario Discovery," *Technological Forecasting & Social Change* 77 (2010): 34–49, doi:10.1016/j. techfore.2009.08.002.
- J.W. Forrester, Industrial Dynamics (Cambridge, Massachusetts: MIT Press, 1961); John D. Sterman, Business Dynamics: Systems Thinking and Modeling for a Complex World (Boston: McGraw-Hill, 2000).
- Naomi Oreskes, Kristin Shrader-Frechette, and Kenneth Belitz, "Verification, Validation, and Confirmation of Numerical Models in the Earth Sciences," *Science* 263, no. 5147 (1994): 641–46, doi:10.1126/ science.263.5147.641.
- Philip N. Johnson-Laird, *Mental Models: Towards a Cognitive Science of Language, Inference, and Consciousness.* (Cambridge, MA: Harvard University Press, 1983).
- Robert J. Lempert, Steven W. Popper, and Steven C. Bankes, "Shaping the next One Hundred Years : New Methods for Quantitative, Long-Term Policy Analysis" (Santa Monica: RAND, 2003), http://www.rand.org/pubs/ monograph_reports/2007/MR1626.pdf.
- Robert J. Lempert et al., "A General, Analytic Method for Generating Robust Strategies and Narrative Scenarios," Management Science 52, no. 4 (2006): 514–28, doi:10.1287/mnsc.1050.0472.
- 25. Forrester, Industrial Dynamics; Sterman, Business Dynamics: Systems Thinking and Modelling for a Complex World.
- 26. Lempert, Popper, and Bankes, "Shaping the next One Hundred Years : New Methods for Quantitative, Long-Term Policy Analysis."
- 27. Paul Collier and Anke Hoeffler, "Greed and Grievance in Civil Wars," Oxford Economic Papers 56 (2004): 563–95.
- Auping, Willem L. et al., "Simulating the Impact of Climate Mitigation Policies on Social Unrest in Rentier States" (34th International Conference of the System Dynamics Society, Delft, Netherlands: System Dynamics Society, 2016).

- 29. Auping, Willem L. et al., "The Geopolitical Impact of the Shale Revolution: Exploring Consequences on Energy Prices and Rentier States," Energy Policy (Netherlands, 2016); Sijbren de Jong, Willem Lucas Auping, and Joris Govers, "The Geopolitics of Shale Gas. The Implications of the US' Shale Gas Revolution on Intrastate Stability within Traditional Oil and Natural Gas Exporting Countries in The EU Neighborhood," Strategy & Change (The Hague: The Hague Centre for Strategic Studies, 2014).
- 30. Jong, Auping, and Govers, "The Geopolitics of Shale Gas."
- "Oil Rents (% of GDP) | Data | Table," World Bank, 2012, http://data.worldbank.org/indicator/NY.GDP.PETR.RT.ZS;
 "Natural Gas Rents (% of GDP) | Data," World Bank, accessed December 22, 2016, http://data.worldbank.org/ indicator/NY.GDP.NGAS.RT.ZS; "Polity Project," The Center for Systemic Peace, 2016, http://www.systemicpeace. org/polityproject.html; "U.S. Energy Information Administration (EIA) Country Reports," 2016, http://www.eia.gov/ beta/international/.
- 32. World Bank, Trading Economics, SWF Institute, Deutsche Bank.
- See for example "Saudi Arabia Foreign Exchange Reserves | 2010-2016 | Data | Chart," *Trading Economics*, 2016, http://www.tradingeconomics.com/saudi-arabia/foreign-exchange-reserves.
- 34. "Polity IV Project: Home Page," *Center for Systemic Peace*, June 10, 2013, http://www.systemicpeace.org/polity/polity4.htm.
- 35. Trading Economics, World Bank.
- The World Bank, "Unemployment, Youth Total (% of Total Labor Force Ages 15-24) (Modeled ILO Estimate)," 2016, http://data.worldbank.org/indicator/SL.UEM.1524.ZS.
- 37. The conversion of data was based *inter alia* on the following sources: "Guidelines to Renewable Energies" (People's Democratic Republic of Algeria, Ministry of Energy and Mines, n.d.); "The Prodigal Sun: Renewables Have Huge Potential for Domestic Use, and Possibly Export. Algeria | Energy.," *Oxford Business Group*, December 8, 2013, http://www.oxfordbusinessgroup.com/analysis/prodigal-sun-renewables-have-huge-potential-domestic-use-and-possibly-export; "Renewable Energy Snapshot: Azerbaijan" (United Nations Development Programme (UNDP), 2013); "Renewable Energy Snapshot: Kazakhstan" (United Nations Development Programme (UNDP), 2013); P Govinda Rao, "Assessment of Solar Energy Potential in Qatar," November 2013; Iskander Tlili, "Renewable Energy in Saudi Arabia: Current Status and Future Potentials," *Environment, Development and Sustainability: A Multidisciplinary Approach to the Theory and Practice of Sustainable Development* 17, no. 4 (2015): 859–86; Hakim Saibi, "Geothermal Resources in Algeria," *Renewable and Sustainable Energy Reviews* 13, no. 9 (December 2009): 2544–52, doi:10.1016/j.rser.2009.06.019; Kamel Abdeladim et al., "RENEWABLE ENERGIES IN ALGERIA: CURRENT SITUATION AND PERSPECTIVES," in *ResearchGate*, 2014, doi:10.13140/2.1.1600.4808.
- OECD, "Biodiesel Production, OECD Countries, 1990-2011," OECD Publishing, 2013, http://www.keepeek.com/ Digital-Asset-Management/oecd/agriculture-and-food/oecd-compendium-of-agri-environmental-indicators/ biodiesel-production-oecd-countries-1990-2011_9789264186217-table33-en#page1.
- FAO et al., "Price Volatility in Food and Agricultural Markets: Policy Responses," June 2, 2011, http://www.oecd. org/trade/agricultural-trade/48152638.pdf.

- Koen Overmars et al., "Estimate of Indirect Land Use Change from Biofuels Based on Historical Data," JRC Science and Policy Briefs (European Commission, n.d.), 9.
- Donald Mitchell, "A Note on Rising Food Prices," Policy research working paper (The World Bank, July 2008), http://www-wds.worldbank.org/servlet/WDSContentServer/WDSP/IB/2008/07/28/000020439_2008072810300 2/Rendered/PDF/WP4682.pdf.
- 42. De Jong, Auping, and Govers, "The Geopolitics of Shale Gas."
- 43. IMF data.
- 44. Iran stands out economically due to the removal of sanctions and the expansion in oil production that partly compensated a decline in oil price.
- 45. Martin Sommer et al., "Learning to Live with Cheaper Oil Policy Adjustment in Oil-Exporting Countries of the Middle East and Central Asia" (International Monetary Fund Middle East and Central Asia Department, 2016), http://www.imf.org/external/pubs/ft/dp/2016/mcd1603.pdf.
- 46. "Vision 2030" (Kingdom of Saudi Arabia, 2016).
- 47. Ibid., 47.
- 48. Ibid., 7, 39, 49, 51 and 61.
- Simon Kerr, "Saudi Arabia Seeks \$30bn-\$50bn Solar and Wind Energy Investment," *Financial Times*, January 16, 2017, https://www.ft.com/content/d370829e-dbfe-11e6-86ac-f253db7791c6.
- 50. "Saudi Arabia's Post-Oil Future," *The Economist*, April 2016, http://www.economist.com/news/middle-east-andafrica/21697673-bold-promises-bold-young-prince-they-will-be-hard-keep-saudi-arabias.
- 51. "Distance to Frontier Doing Business World Bank Group," *The World Bank*, 2016, http://www.doingbusiness. org/data/distance-to-frontier.
- 52. Data from SIPRI.
- International Renewable Energy Agency, "Renewable Energy Country Profiles Africa" (International Renewable Energy Agency, 2011), http://www.irena.org/DocumentDownloads/Publications/CountryProfiles_Africa_WEB.pdf.
- African Development Bank Group, "Renewable Energy in Africa- Mali Country Profile," Renewable Energy in Africa (African Development Bank Group, 2015), http://www.afdb.org/fileadmin/uploads/afdb/Documents/ Generic-Documents/Profil_ER_Mal_Web_light.pdf.
- 55. "The World Factbook Central Intelligence Agency Ethiopia," January 12, 2017, https://www.cia.gov/library/ publications/resources/the-world-factbook/geos/et.html.
- International Energy Agency, "IEA Report- Senegal," International Energy Agency, 2014, https://www.iea.org/ statistics/statisticssearch/report/?country=Senegal&product=electricityandheat.
- 57. International Renewable Energy Agency, "Renewable Energy Country Profiles Africa."
- 58. Ibid.
- 59. Antonio Castellano et al., "Brighter Africa," *McKinsey*, February 2015, http://www.mckinsey.com/industries/ electric-power-and-natural-gas/our-insights/powering-africa.
- 60. "Pros and Cons of Hydroelectric Power," *Conserve Energy Future*, February 17, 2015, http://www.conserve-energy-future.com/pros-and-cons-of-hydroelectric-power.php; "Drought: Hydropower's Achilles Heel," *International Rivers*, May 16, 2016, https://www.internationalrivers.org/blogs/433/drought-hydropower-s-achilles-heel.

- 61. Castellano et al., "Brighter Africa," 22.
- 62. International Renewable Energy Agency, "Africa 2030: Roadmap for a Renewable Energy Future," 2015, 22, http:// www.irena.org/menu/index.aspx?mnu=Subcat&PriMenuID=36&CatID=141&SubcatID=641.
- International Energy Agency, "Africa Energy Outlook," International Energy Agency, 2014, https://www.iea.org/ publications/freepublications/publication/weo-2014-special-report-africa-energy-outlook.html.
- 64. Ibid., 37.
- 65. Ibid., 77.
- Chineme Okafor, "Nigeria's Leap Towards Solar Energy," *This Day (Lagos)*, July 19, 2016, http://allafrica.com/ stories/201607190987.html.
- 67. "Power Africa in Ethiopia," USAID, September 28, 2016, https://www.usaid.gov/powerafrica/ethiopia.
- Tom Kenning, "Kenya to Construct 55MW Solar Plant with Chinese Government Funding," *PV-Tech*, April 4, 2016, http://www.pv-tech.org/news/kenya-to-construct-55mw-solar-plant-with-chinese-government-funding.
- 69. Nick Butler, "China's Energy Strategy: Power and Independence," *Financial Times*, January 16, 2017, http://blogs. ft.com/nick-butler/2017/01/16/chinas-energy-strategy-power-and-independence/.
- 70. Tim Buckley and Simon Nicholas, "China's Global Renewable Expansion. How the Worl's Second-Biggest National Economy Is Positioned to Lead the World in Clean-Power Investment" (Institute for Energy Economics and Financial Analysis, January 2017), 1; Joel Jaeger, Paul Joffe, and Ranping Song, "China Is Leaving the U.S. Behind on Clean Energy Investment | World Resources Institute," January 6, 2017, http://www.wri.org/blog/2017/01/ china-leaving-us-behind-clean-energy-investment.
- 71. International Energy Agency, "Boosting the Power Sector in Sub-Saharan Africa- China's Involvement," Partner Countries Series (International Energy Agency, 2016), https://www.iea.org/publications/freepublications/ publication/Partner_Country_SeriesChinaBoosting_the_Power_Sector_in_SubSaharan_Africa_Chinas_ Involvement.pdf.
- 72. Buckley and Nicholas, "China's Global Renewable Expansion. How the Worl's Second-Biggest National Economy Is Positioned to Lead the World in Clean-Power Investment," 7.
- 73. Castellano et al., "Brighter Africa," 33.
- Amadou Oury Diallo, "Mali: un projet de centrale solaire de 50MW a Kita," Jeune Afrique, November 5, 2015, https://www.akuoenergy.com/fileadmin/media/pdf/press/2015-11-05-5959@WWW_JEUNEAFRIQUE_COM.pdf.
- International Renewable Energy Agency, "Solar PV in Africa: Costs and Markets," 30, accessed October 6, 2016, http://www.irena.org/menu/index.aspx?mnu=Subcat&PriMenulD=36&CatID=141&SubcatID=2744.
- Amadou Ndiaye, "Le Sénégal se dote de sa première centrale solaire à Bokhol," *Le Monde*, March 23, 2016, http://www.lemonde.fr/afrique/article/2016/03/23/le-senegal-se-dote-de-sa-premiere-centrale-solaire-abokhol_4888448_3212.html.
- Gaëlle Laleix, "La plus grande centrale solaire d'Afrique de l'Ouest a trouvé ses financements," *RFI*, May 17, 2016, http://www.rfi.fr/emission/20160517-plus-grande-centrale-solaire-afrique-ouest-trouve-financements.
- "Renewable Energy: The Case of Botswana," Text/HTML, World Bank, accessed September 29, 2016, http:// www.worldbank.org/en/news/feature/2014/08/11/renewable-energy-the-case-of-botswana.

- "Africa's Largest Microgrid Takes Shape In Equatorial Guinea | Engerati Generation," *Engerati*, December 16, 2014, https://www.engerati.com/article/africa%E2%80%99s-largest-microgrid-takes-shape-equatorial-guinea.
- 80. Selwa Calderbank, "Desertec Abandons Sahara Solar Power Export Dream," *EurActiv*, May 31, 2013, http://www. euractiv.com/section/trade-society/news/desertec-abandons-sahara-solar-power-export-dream/.
- Republic of Angola Ministry of Energy and Water, "The Renewable Energies in Angola Current Picture and Perspectives," n.d., http://www.energyafrica.de/fileadmin/user_upload/Energy-Africa__12/Presentation_ Ministry%20of%20Energy%20and%20Water_The%20renewable%20energies%20in%20Angola.pdf.
- Renewable Energy Policy Network for the 21st Century, "Renewables 2016 Global Status Report"; "Lake Turkana Wind Power | Project Overview," 2017, http://ltwp.co.ke/.
- 83. International Renewable Energy Agency, "Africa 2030: Roadmap for a Renewable Energy Future," 1, 26.
- 84. International Renewable Energy Agency, "Africa 2030: Roadmap for a Renewable Energy Future."
- 85. International Energy Agency, "Africa Energy Outlook," 59.
- 86. Ibid., 37.
- K. Stecher, A. Brosowski, and D. Thrän, "Biomass Potential in Africa" (IRENA and DBFZ, 2013), 14, http://www. irena.org/menu/index.aspx?mnu=Subcat&PriMenuID=36&CatID=141&SubcatID=349.
- Matthew Owen, Robert van der Plas, and Steve Sepp, "Can There Be Energy Policy in Sub-Saharan Africa without Biomass?," *Energy for Sustainable Development*, Special Issue on Charcoal, 17, no. 2 (April 2013): 147, doi:10.1016/j.esd.2012.10.005.
- 89. Stecher, Brosowski, and Thrän, "Biomass Potential in Africa," 13.
- 90. International Renewable Energy Agency, "Renewable Energy Country Profiles Africa."
- 91. Ibid.
- 92. Ibid., 39.
- 93. International Renewable Energy Agency, "Renewable Energy Country Profiles Africa."
- 94. Owen, Plas, and Sepp, "Can There Be Energy Policy in Sub-Saharan Africa without Biomass?," 147.
- 95. Ibid.
- 96. Ibid., 148.
- 97. International Energy Agency, "Africa Energy Outlook," 14.
- Ausilio Bauen et al., "Bioenergy- A Sustainable and Reliable Energy Source. A Review of Status and Prospects" (IEA, June 2009), 2, http://www.seai.ie/Renewables/Bioenergy/Bioenergy_%E2%80%93_a_Sustainable_and_ Reliable_Energy_Source_MAIN_REPORT.pdf.
- 99. International Energy Agency, "Africa Energy Outlook."
- 100. Ibid., 77.
- 101. Energy Transition Advisors, "Carbon Tracker Initiative," Energy Access (ETA, November 2014), https://www.carbontracker.org/wp-content/uploads/2014/11/Coal-Energy-Access-111014-final.pdf.
- "The World Factbook CAR Central Intelligence Agency," accessed October 5, 2016, https://www.cia.gov/ library/publications/the-world-factbook/geos/ct.html.
- See International Energy Agency, "World Energy Outlook Energy Access Database," Database, International Energy Agency, (2016), http://www.worldenergyoutlook.org/resources/energydevelopment/ energyaccessdatabase/.

104. Castellano et al., "Brighter Africa."

```
105. Ibid., 13.
```

- 106. "Private Participation," Angola Energy 2025- Angola Power Sector Long Term Vision, accessed October 26, 2016, http://www.angolaenergia2025.com/en/conteudo/private-participation.
- 107. Angolan Ministry of Energy and Water, "RURAL ELECTRIFICATION," *Republica de Angola Ministerio Da Energia E Aguas*, 2016, http://www.angolaenergia2025.com/en/conteudo/rural-electrification.
- CAR Ministry of Mines Energy and Hydraulics, "Expression of Interest by the Central African Republic to CIF/SREP Program" (World Bank Group, 2014), 6, 17-10-2016, https://www-cif.climateinvestmentfunds.org/sites/default/ files/meeting-documents/car_eoi_0.pdf.
- See e.g. World Bank, "Off-Grid Solar Lighting Up Ethiopia," World Bank, August 15, 2016, http://www.worldbank. org/en/news/feature/2016/08/15/off-grid-solar-lighting-up-ethiopia.
- 110. International Energy Agency, "Africa Energy Outlook," 14.
- 111. Deutsche Gesellschaft für Internationale Zusammenarbeit, "The Nigerian Energy Sector- An Overview with a Special Emphasis on Renewable Energy, Energy Efficiency and Rural Electrification" (Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ), June 2015), 19, https://www.giz.de/en/downloads/giz2015-en-nigerianenergy-sector.pdf.
- 112. "Botswana 10th National Development Plan 2009-2016," 2009, 240, http://www.nationalplanningcycles.org/ sites/default/files/country_docs/Botswana/ndp_botswana.pdf.
- 113. "Angola- U.S. Energy Information Administration (EIA)," *Marcon*, accessed October 17, 2016, http://www.marcon. com/marcon2c.cfm?SectionListsID=30&PageID=2262.
- 114. Deutsche Gesellschaft für Internationale Zusammenarbeit, "The Nigerian Energy Sector."
- 115. International Renewable Energy Agency, "Africa 2030: Roadmap for a Renewable Energy Future."
- KPMG Africa Infrastructure & Major Projects Gropu, "Sub-Saharan Africa Power Outlook 2014" (KPMG, 2014), http://www.kpmg.com/ZA/en/IssuesAndInsights/ArticlesPublications/General-Industries-Publications/ Documents/2014%20Sub-Saharan%20Africa%20Power%20Outlook.pdf.
- 117. Federal Democratic Republic of Ethiopia Ministry of Water and Energy, "Scaling-up Renewable Energy Program Ethiopia Investment Plan (Draft Final)" (OECD, January 2012), 5, https://www.oecd.org/env/cc/TADELE_FDRE%20 Ethiopia%20Scaling%20-%20Up%20Renewable%20Energy%20Program%202012.pdf.
- 118. International Energy Agency, "Africa Energy Outlook," 69.
- World Bank Group, "Sub-Saharan Africa," Global Economic Prospects (World Bank Group, January 2016), 156, https://www.worldbank.org/content/dam/Worldbank/GEP/GEP2016a/Global-Economic-Prospects-January-2016-Sub-Saharan-Africa-analysis.pdf.
- Maggie Fick, "Oil and Gas Discoveries Fuel Senegalese Fears," *Financial Times*, December 27, 2016, https:// www.ft.com/content/065ba414-bbce-11e6-8b45-b8b81dd5d080.
- 121. World Bank Group, "Sub-Saharan Africa," 156.
- Oxford Business Group, "Gabon Aspires to Emerging Market Status," Oxford Business Group, August 18, 2015, https://www.oxfordbusinessgroup.com/analysis/reviewing-plan-progress-towards-goal-reaching-emergingmarket-status.
- 123. South African Government, "Energy," *South African Government*, accessed October 25, 2016, http://www.gov.za/ about-SA/energy.

- 124. See "CVF Participating Countries," *Climate Vulnerable Forum*, accessed November 18, 2016, http://www.thecvf. org/web/climate-vulnerable-forum/cvf-participating-countries/; Axel Bojanowski, "Kampf Gegen Klimawandel: 45 Länder Wollen Komplett Auf Kohle, Öl Und Gas Verzichten," *Spiegel Online*, November 18, 2016, http://www. spiegel.de/wissenschaft/natur/klimakonferenz-in-marrakesch-45-laender-verzichten-komplett-auf-kohle-oel-undgas-a-1121838.html.
- 125. The Global Commission on Economy and Climate, "The Sustainable Infrastructure Imperative: Financing for Better Growth and Development. The 2016 New Climate Change Economy Report," The New Climate Economy (Washington/London: The Global Commission on the Economy and Climate, 2016), http://newclimateeconomy. report/2016/wp-content/uploads/sites/4/2014/08/NCE_2016Report.pdf.
- 126. See e.g. "Southern African Power Pool," accessed December 23, 2016, http://www.sapp.co.zw/; and "Eastern Africa Power Pool, accessed December 23, 2016, http://eappool.org/.
- 127. Castellano et al., "Brighter Africa."
- 128. The Global Commission on Economy and Climate, "2016 NCE Report," 55.
- 129. Ibid., 48.
- "Energieagenda. Naar Een CO₂-Arme Energievoorziening" (Den Haag: Ministerie van Economische Zaken, December 2016), 86.

The Hague Centre for Strategic Studies

Lange Voorhout 16 2514 EE The Hague The Netherlands info@hcss.nl HCSS.NL